

Received 21 April 2022, accepted 31 May 2022, date of publication 21 June 2022, date of current version 1 July 2022. *Digital Object Identifier* 10.1109/ACCESS.2022.3184704

Blockchain for Distributed Energy Resources Management and Integration

SERGIO CANTILLO-LUNA^{®1}, (Student Member, IEEE), RICARDO MORENO-CHUQUEN¹, (Member, IEEE), HAROLD R. CHAMORRO^{®2}, (Senior Member, IEEE), VIJAY K. SOOD³, SHAHRIAR BADSHA⁴, (Member, IEEE), AND CHARALAMBOS KONSTANTINOU^{®5}, (Senior Member, IEEE)

¹Faculty of Engineering, Universidad Autónoma de Occidente, Cali 760030, Colombia

²Power Systems and Computer Science Department, KTH Royal Institute of Technology, 10044 Stockholm, Sweden

³Department of Electrical, Computer and Software Engineering, Ontario Tech University, Oshawa, ON L1H7K4, Canada

⁵Computer, Electrical and Mathematical Sciences and Engineering Division, King Abdullah University of Science and Technology (KAUST), Thuwal 23955, Saudi Arabia

Corresponding author: Sergio Cantillo-Luna (sergio.cantillo@uao.edu.co)

ABSTRACT Power grids all over the world are transitioning towards a decentralized structure. Under such a transition, blockchain technology is emerging as a potential solution for technical, deployment and decentralization issues, given its security, integrity, decentralized nature and required infrastructure. Moreover, blockchain technology offers excellent features like non-repudiation and immutability which makes it a promising application for DER integration and management on reliability factors. In this paper, a comprehensive review of blockchain applications for DER management and integration is presented. First, a blockchain-based literature review of research activities in the DER integration area and related tasks including entrepreneurial efforts is carried out. Next, the different opportunities and challenges of DER integration and management in power grids, i.e., centralization, regulatory support, development costs are discussed. Finally, some key research challenges and opportunities of including blockchain technology to DER integration and management issues are presented.

INDEX TERMS Blockchain, distributed energy resources (DER), distributed ledger technologies, consensus algorithms.

NOTATION

This section presents the main notations used in the document.

AI	Artificial Intelligence.						
AMI	Advanced Metering Interfaces.						
DER	Distributed Energy Resources.						
DERMS	Distributed Energy Resources Management						
	System.						
DG	Distributed Generation.						
DLT	Distributed Ledger Technologies.						
DPoS	Delegated Proof-of-Stake.						
DR	Demand Response.						
DSO	Distribution System Operator.						

The associate editor coordinating the review of this manuscript and approving it for publication was Bin Zhou^(D).

ESS	Energy Storage Systems.
EV	Electric Vehicles.
HEMS	Home Energy Management System.
IoT	Internet of Things.
IoE	Internet of Energy.
ICT	Information and Communication Technology.
LCOE	Levelized Cost of Energy.
OPF	Optimal Power Flow.
PoA	Proof-of-Authority.
PoC	Proof-of-Capacity.
PBFT	Practical Byzantine Fault Tolerance.
PoS	Proof-of-Stake.
PoW	Proof-of-Work.
PV	Photovoltaics.
P2P	Peer-to-Peer.
SGs	Smart grids.

⁴BOSCH Engineering, Farmington Hills, MI 48331, USA

VPPs Virtual Power Plants.

V2V Vehicle-to-Vehicle.

V2G Vehicle-to-Grid.

I. INTRODUCTION

The transition to sustainable energy is encouraging the integration of increasing amounts of renewable energy (i.e., wind and solar PV) into the electrical grids around the world [1], [2]. The development of renewable energy sources (RES) has been enabled by the deregulation of the energy sector and driven by the promise of financial rewards and energy policy initiatives in several countries. These situations have led to an energy transition trend towards decarbonization. decentralization, digitalization and democratization. Thus, the traditional way of generating, transmitting and distributing electricity has changed significantly as generation units are now more distributed, more efficient and closer to the end-users [3]–[6] offering them opportunities for individual participation and empowerment [7], [8].

This process is bringing meaningful changes as traditional large-scale synchronous generators are being replaced by emerging small-scale distribution-level low-carbon technologies (i.e., local distributed generation (DG) in cities, energy storage systems (ESS), electric vehicles (EVs), and demand response (DR)) [9], [10]. All of these resources are lumped together under the label of *distributed energy resources* (DER). Many investments have been made worldwide to develop more affordable, flexible and efficient solutions that allow a wider adoption of these resources at the end-user level for improving the efficiency, reliability and flexibility of electric grids [3], [11].

The intended benefits of efficient and progressive DER integration and management are considerable [12]–[17]. Endusers such as residences or industrial users are empowered by the inherent DER flexibility, since they are now able to manage their energy production and consumption. Now, owners can participate actively in the energy acquisition and cost reduction by doing energy management. Moreover, DER flexibility is valuable for power systems, as they offer opportunities to alleviate grid infrastructure investment costs, and provide ancillary services such as frequency control, voltage control and storage reserves, amongst others.

It is important to mention that decentralization in power systems using DER can occur in several dimensions: physical decentralization, which refers to the disaggregation of power generation through the installation of small or medium size energy resources directly to low or medium voltage distribution grids [18], [19]; market decentralization, meaning new trading and clearing mechanisms that allow financial and data integration to DERs in a secure, local and straightforward way with end-users (e.g., P2P energy trading); and decentralization of the operation, where more than one *manager* (e.g., DERMS, VPP, Local Energy Communities, among others) can transact selling and buying of energy in order to

clear locally its energy supply and demand [20]. Hence, this multidimensional and massive process brings with it some challenges to be overcome.

Some of the most important challenges are as follows: mitigating uncertainty of renewables generation, reduction of high and fluctuating energy prices for end-users [16], managing in real-time supply and demand energy gaps [21], avoiding grid instability and low power quality scenarios [22], alleviating distribution grid needs with profits from new players in energy markets, amongst others.

Blockchain technology offers a new alternative to overcome these DER challenges for several reasons: It provides a trusted environment for different stakeholders to interact peer-to-peer with enhanced security for energy, data, and financial transactions. Blockchain technology can also provide faster and smoother decentralized system operation. Moreover, it is inherently a decentralized mechanism that works in conjunction with other energy-related technologies and paradigms such as Smart Grids (SGs) [23], [24], peer-topeer energy trading [25]–[28], virtual power plants [29]–[32], IoT [33], [34], among others, looking to enhance their features wherever a coordinated integration (technical, market, and financial) of these resources is sought, in order to participate in energy markets, providing value and empowerment to consumers and offering different grid services due to their security, reliability, and efficiency in DER-integrated power systems.

A. KEY CONTRIBUTIONS

It is important to mention that several studies have been conducted on the influence of blockchain technology on distributed energy resources (DER) and their applications. However, these studies have been oriented to specific and limited scientific contexts (i.e., reduced location, pilot projects) or have not addressed other initiatives around this topic due to its complexity and novelty.

This expresses the need for a study that comprehensively shows the efforts made for the blockchain technology use in DER management and integration by the different stakeholders (i.e. scientific community, business initiatives, among others), as well as analyzing the challenges that still persist for this approach. Table 1 summarizes important survey articles related to the application of blockchain technology in different areas related with DER management and integration. The terms high, moderate and low are the levels of coverage and complexity in each presented area.

The review articles presented in this table show the previously mentioned, the big interest in showing different blockchain applications for DER management and integration as well as their current barriers and future challenges according to the time where they have been published, many of them related with exploratory analyses of a mostly technical nature and focused on development pilots. However, it is also evident that these studies have not focused on showing business efforts or applications in this same line (e.g. crowdfunding), or a further study in the key issues at

TABLE 1. Summary of important surveys on blockchain for DER management and integration.

Reference	Blockchain Integration for DER	Coverage of blockchain applications	Business Initiatives	Technical Aspects	Multi- dimensional	Future Challenges	Remarks
[35]	High	High	High	Low	Moderate	High	It comprehensively covers blockchain-based applications of different DERs, as well as re- lated future challenges.
[36]	Moderate	Moderate	Low	High	High	High	It specifies the future direction and challenges of blockchain application in SG thought to- wards the security of integrating DER.
[37]	High	Moderate	Low	Moderate	Moderate	Low	Concise details of blockchain applications re- lated with DER management are provided.
[38]	Moderate	Moderate	Low	Low	Moderate	Moderate	It specifically details the multidimensional challenges of blockchain applied to the energy sector.
[39]	Moderate	Moderate	Low	Moderate	Low	High	The development of blockchain in the energy internet paradigm is reviewed and analyzed (challenges, issues and features).
[40]	Moderate	Moderate	Low	Moderate	Moderate	Moderate	Specifies the use and future issues and chal- lenges of blockchain technology in the energy internet paradigm.
[41]	Moderate	Moderate	Low	Moderate	Low	Low	Details various applications of blockchain technology used on different types of DER.
[42]	High	Moderate	Moderate	Moderate	High	High	Describes the applicability of blockchains specifically on SGs. Clearly specifies current limitations and future development trends.
[43]	High	High	Low	High	High	Moderate	Different DER integration approaches and their respective considerations (limitations, formulations, among others) are analyzed.
[44]	Moderate	High	Low	Moderate	Low	Moderate	Blockchain-based applications for different types of DER are specified, as well as evalu- ating different architectures to consider other potential uses.
[45]	Moderate	High	High	Moderate	High	Moderate	The challenges faced by current solutions are identified, and the frameworks and techniques used to integrate blockchain are highlighted.
[46]	Moderate	High	Moderate	Moderate	High	High	The limitations of blockchain and its impact on energy systems are analyzed in depth. They also show an overview of different blockchain- based applications.

other dimensions for the Blockchain development for these tasks.

Therefore, this review paper offers insights on up-to-date activities of rapidly growing blockchain applications focused on DER integration and management from different perspectives (technical, regulatory, economic, among others), highlighting recent challenges and opportunities. Furthermore, this paper also presents some applications in different areas where blockchain technology has a great influence on DER management and integration tasks. The key contributions of this paper are summarized as follows:

- A description of blockchain technology strategies (e.g., categories, consensus mechanisms, smart contracts, etc.) related with DER integration and management.
- The current research scope of blockchain related technologies defined by DER management and integration applications including entrepreneurial efforts.
- A review of different issues and development opportunities related to blockchain implementation for DER.

This paper is organized as follows: Section II provides some theoretical background of blockchain technology as it relates to DER integration and management. Section III presents different paradigms of DER management and integration where Blockchain has contributed significantly, as well as the role of ICT in this process. Section IV addresses the different areas and applications of blockchain technology for DER. Section V highlights the challenges and opportunities that blockchain technology presents in DER management and integration. Section VI discusses the effects of blockchain technology implementation on DER applications. Finally, concluding remarks are provided in Section VII.

II. BLOCKCHAIN TECHNOLOGY

This section presents the blockchain technology as a support for DER integration and management. Three topics are included: distributed ledger technology, blockchain and smart contracts, and the consensus mechanisms.

A. DISTRIBUTED LEDGER TECHNOLOGY (DLT)

Like a database, the electronic distributed ledger technology is defined for recording information that is not executed by a single entity (i.e., a centralized ledger). These DLTs allow



FIGURE 1. Comparison between centralized and distributed ledger technologies.

data storage and use in a decentralized fashion with peers and distributed (connected and therefore can communicate) in a private or public way, as shown in Figure 1. Any modification made in the DLTs is shared and validated by all the nodes (consensus or agreement) within seconds or minutes in order to offer security, transparency and reliability [36], [47], [48]. Blockchain is a type of DLT which has the capability to empower different parties take on an active role in energy transactions by eliminating third-party involvement, so that consumers, prosumers and producers can interact directly [49].

B. BLOCKCHAIN AND SMART CONTRACTS

A *blockchain* is defined as a distributed registration system that promotes decentralization (actors or blocks are connected), transparency (does not require intermediaries) and data integrity (verified by all nodes in the network). Authors in [35], [47], [50]–[54] define it more specifically as a secure (immutable, append-only) digital data structure that runs on digital networks that contain an information database with different transactions, records, or files in chronological order, where many peers can participate. Also, other authors in [36], [55], [56] add that it is a collection of information "blocks" (e.g., hashes, scripts, timestamps, transactions, amongst others) distributed in chains (especially by cryptography), which when created, are distributed to all nodes that they share.

Based on information accessibility, blockchain systems can be categorized into private (partially decentralized systems with high-efficiency, where some selected and trusted users are permitted to read, validate and participate in publishing new blocks to the system [36], [57], [58]), public (fully decentralized systems with lower efficiency compared to private ones, require mandatory consensus mechanisms for security and data integrity, allow open participation by providing a copy of the blockchain to all its members, and it is almost impossible to tamper [59]) and consortium blockchains (a special type of blockchain because it is semidecentralized, where predefined groups of nodes that can validate transactions through consensus mechanism in order to make transactions faster [60]). All of these blockchain topologies can be used to integrate and manage DER depending on the context and task involved.

The *Smart Contracts*, on the other hand, are essential for current blockchain ecosystems since they are scripts written

with programming languages that can self-execute actions in an immutable, transparent and completely safe way according to a series of parameters already established (in most cases solidity). Other authors such as [61]–[63] define it more precisely as an auto-executable and immutable script that verifies specific actions like contractual clauses in a regular contract. When certain conditions are fulfilled corresponding actions will be executed.

As with DLT, smart contracts can help eliminate the brokerage part in various use cases, reducing transaction costs and allowing for low-value transactions. A smart contract is a piece of code stored and deployed in a blockchain without a control entity in the blockchain framework. Instead of legal terms, these contracts store the conditions and events such as a value or a deadline date or simply transaction information [36], [64]. In this way, other authors [65] agree that smart contracts are most efficient for contracts that can be reduced to simple "if-then" statements, as their terms are easy to convert to computer code and can be executed automatically. Based on [66], Figure 2 shows the life cycle of a smart contract.



FIGURE 2. Life cycle of a smart contract.

C. CONSENSUS MECHANISMS

The consensus mechanisms or algorithms are key components for the proper blockchain operation, since they are responsible for determining the correct record state after making a transaction between nodes. However, these mechanisms are not universal for all blockchain due to their performance and privacy features. Based on [67], Figure 3 shows, how any blockchain works with a consensus mechanism. Next, some of the most popular consensus mechanisms applied in DER management and integration are explained.

1) PROOF OF WORK (PoW)

The Proof-of-Work is the most common consensus algorithm that ensures a record (i.e., the name used for the set of blocks that store all operations) is immutable (too expensive to convince the 51% of blockchain nodes to recalculate). Also, authors in [36], [68]–[70] extend this meaning by including the main objective of PoW when creating a new block, is to solve a computationally costly cryptographic puzzle with computing power which is challenging to solve but easy to verify (*mining*). After this process, the first node to solve this puzzle receives a reward (*fee*) as compensation. Once mined and paid for, the solution is attached to the new block and transmitted over the network.



FIGURE 3. How the blockchain with validation and consensus mechanism works.

2) PROOF OF STAKE (PoS)

The Proof-of-Stake is the most popular consensus mechanism or algorithm alternative to PoW. This method is oriented to solve PoW issues related to scalability and security. The algorithm randomly selects the nodes that will validate the creation of new blocks. The probability of a node being chosen is proportional to the assets balance (or longevity) that the node has. Other authors [36], [64], [69]–[73] indicate that the algorithm requires far fewer CPU computations and energy for mining instead of running a high computational puzzle-solving effort that PoW involves.

3) DELEGATED PROOF OF STAKE (DPoS)

This is a PoS consensus mechanism variant, which is designed for a highly scalable blockchain. The implementation of this protocol offers Byzantine Fault Tolerance (BFT). It provides high levels of security in public blockchains (that is highly desirable in decentralized energy systems). The mechanism of DPoS is based on all the network peers vote for some "delegates" nodes, which rotate amongst themselves so that each one can produce a block and collect a reward for it [72], [74], [75]. It provides the network with lower power consumption, faster transaction creation and validation, and high tolerance to tampering attempts [76].

4) PROOF OF AUTHORITY (PoA)

The Proof-of-Authority is another BFT-based consensus algorithm that is designed as a practical and efficient solution (lower energy consumption compared to PoW and PoS) [23], especially aimed at private and permissioned blockchains (i.e., limited participation, and restricted rights to validate transactions with fault tolerance). PoA takes advantage of real identities to allow validation within a blockchain system. That means, the authenticators (called *authorities* in this algorithm) put their real identity and reputation as transparency guarantee in order to distribute the responsibility of creating new blocks [77].

5) PRACTICAL BYZANTINE FAULT TOLERANCE (PBFT)

The Practical Byzantine Fault tolerance is a consensus mechanism dedicated to replicating state machines that tolerate the presence of blockchain malicious nodes (i.e., independent node failures or attacker nodes in the network sending out misleading information). Essentially, these nodes communicate with each other with ordering, where all honest nodes (at least two-thirds or 66% or 3f + 1 nodes, where f is the number of faulty nodes or data) in the network agreed (votingbased) on the blockchain status (creation and reception of blocks) [40], [69], [78].

6) PROOF OF CAPACITY (PoC)

The PoC algorithm (also known as Proof-of-Space) is a particular consensus mechanism since it allows miners to use available hard disk space instead of computational power (PoW) or coins number (PoS), for the transaction deployment (divided in two phases: hard disk plotting with Shabal instead of SHA-256 as encryption algorithm, and mining) [79], [80]. Thus, this consensus mechanism emerged as an alternative to the high energy consumption of PoW as well as the PoS crypto-assets maintenance.

In order to display similarities and differences between these algorithms and show some examples of DER management and integration applications or projects that implement them, a comparison of the above discussed consensus algorithms - based on [81] - is presented in Table 2.

7) IMPACT OF CONSENSUS MECHANISMS ON DER INTEGRATION AND MANAGEMENT

Driven by advances in blockchain technology and the widespread use of DER, different development teams of

TABLE 2.	Consensus	algorithms in	DER management	and integration.
----------	-----------	---------------	----------------	------------------

FEATURE	PoW	PoS	DPoS	PoA	PBFT	PoC		
Type of blockchain	Permissionless	Both	Both	Both	Permissioned	Both		
Energy consumption	High	Moderate	Low	Low	Moderate	Moderate		
Scalability	Strong	Strong	Strong	Strong	Weak	Strong		
Reward	Yes	Yes	Yes	Yes No		Yes		
Hardware	Yes	No	No	No	No	Yes		
dependency								
Computational cost	High	Moderate	Low	Moderate	High	Moderate		
Transaction fees	Yes	Yes	Yes	Yes	No	Yes		
Block creation speed	Low	High	High	High	High	Moderate		
Decentralization	High	Medium	Medium	Low High		Moderate		
Level								
DER management	Power Ledger	SolarCoin [86]	EnLedger [87]	GridSingularity	SunChain [90]	Pylon Network		
and integration	[82]–[85]			[88], [89]		[91]–[93]		
applications/projects	Everty [35], [94]	Energo Labs [95],	Sun Exchange	Slock.it [101]–	LO3 Energy	Chia Network		
		[96]	[97]–[100]	[103]	[104]–[106]	[107], [108]		
	Grid+ [96], [109],	NRGCoin [111]-		Share&Charge	Tennet & Sonnen			
	[110]	[113]		[114]–[116]	[117], [118]			

stakeholders (e.g., utility companies, project pilots, among others) have explored the feasibility of using blockchain technology to manage and integrate these resources from different perspectives [119]–[121]. In this sense, it has been possible to demonstrate the great impact of consensus mechanisms for these tasks.

Firstly, consensus mechanisms with a higher degree of maturity such as PoW and PoS (which are highly scalable, flexible and with low latency) have great influence in digitization and decentralization of DER-associated energy, since they are used in a wide range of applications (e.g., decentralized energy trading, policy or contract terms execution, EV management, among others). These are especially true in the use of devices and enabling technologies for local DER integration and management such as AMI and IoT. However, these mechanisms (especially PoW) present a substantial energy consumption that goes against the objective of sustainability that DERs pursue. Nevertheless, the associated benefits often outweigh the detriments caused by this side effect.

Therefore, other consensus mechanisms (i.e. BFT-based, PoA, PoC, among others) have appeared as an alternative to more specific approaches related to DER management and integration (for enterprise environment, higher security and limited access, lower energy consumption, and taskoriented). However, it also means that in the short-term, there is no unified criteria for the operation of these DER blockchain-based systems, which may negatively impact their coordinated large-scale implementation.

Thus, the consensus mechanisms choice is highly correlated with the use case that has different requirements for performance, latency, security, scalability, environment, privacy, etc. [23], and likewise this indicates the leverage that this will have on the overall performance of the DER-centric system where it is applied.

III. DER INTEGRATION AND MANAGEMENT THROUGH BLOCKCHAIN

Considering the decentralization trend that energy systems currently show due to DER joining and advances in related areas (i.e., market, technical, policies, among others), activities such as the integration and management of these resources have acquired more importance. This situation has led to the development of different approaches that could perform these tasks, according to the current system topology features (voltage levels, leveraging the current infrastructure), the number of potential users and the coordination that could exist between them. Therefore, this section presents under which DER management and integration paradigms, blockchain technology has contributed significantly. On the other hand, the role of ICT in the development of these paradigms is also emphasized.

A. ICT ROLE

As a result of the different advances associated with various cross-cutting trends related to the widespread use of DER (i.e., the 3Ds: decarbonization, digitalization and decentralization trends) coupled with ongoing efforts in different areas (e.g., new devices, systems, and approaches), there has been a deep and seamless integration between energy technology and ICTs, where the latter led by blockchain (or energy blockchain) have delivered features for the proper functioning of decentralized systems that have been developed: privacy, security, transparency, reliability, interoperability, and flexibility. Thus, advanced metering infrastructures as the main exponent of energy-focused ICTs, and due to their capabilities such as enabling real-time communication between consumers and producers, measuring and storing electricity-related data with high resolution, easy combination with other ICT-based solutions (including blockchain), remote management of available energy resources, among others, have become the most important component of this energy-ICT integration [122]. Therefore, the adoption of these ICT-based solutions has led to a significant positive impact on the power systems operation, mainly a more efficient management and control of the available energy resources [123], [124] considering the inclusion of DER as generation assets in energy markets, which highlights the role of these technologies in energy systems road map in the near future [125].

B. DER INTEGRATION AND MANAGEMENT PARADIGMS

As a result of the breadth of the potential benefits of DERs in power systems (i.e., reducing the cost of electricity and grid dependence, providing ancillary services, grid operation upgrading, among others) as well as negative impacts to mitigate (uncertainty, variability, frequency instability scenarios, among others), different paradigms have been developed to integrate and manage them properly. Based on the classification presented in [43], these paradigms fall into three main groups: *individual, coordinated, and decentralized paradigms*.

1) INDIVIDUAL PARADIGMS

The individual paradigms group automated single energy management systems that aim to minimize their cost through the proper management of their resources (i.e., distributed generation, energy storage, demand response) without considering the grid or their neighbors in order to be grouped or coordinated. Examples include mainly home energy management systems (HEMS) and their variations, where service constraints are usually considered.

In this sense, blockchain has mainly contributed as a secure, private and brokerless way of linking these systems to the other paradigms mentioned above, where individual energy management has been combined with different market mechanisms and interaction with other users [19], [126]–[128], in order to mitigate different drawbacks related (e.g., privacy, security, high consumption costs, communication, maximizing self-consumption, among others). Therefore, this technology for this paradigm has not yet had as much relevance compared to the others presented in this paper.

2) COORDINATED PARADIGMS

The coordinated paradigms, on the other hand, present a more cooperative scope, since they have the potential to orchestrate through an intermediary planning entity (i.e., community manager or aggregator) the operation of the DERs of several users for their joint benefit. This joint DER operation is allowed by changes in generation and consumption patterns usually driven by price changes, usually under an optimization problem context. Likewise, this management and integration paradigm is very appealing, due to the different economic incentives offering. Some of the most important representatives of this paradigm are the virtual power plants (VPP), and the different OPF-based aggregation schemes (e.g. local energy communities, microgrids, multi-energy systems, among others).

For this case and unlike the individual paradigm, Blockchain technology has had a greater relevance in this field. This is due to the high compatibility between the topology of this type of network and an interconnected energy scheme. Therefore, the range of areas and applications that have been developed under this paradigm are more extensive: transactive energy (or coordinated and bi-directional energy trading between nodes or users) [129], [130], energy management and scheduling [119], [131]–[133], EV integration [134]–[136], among several others.

3) DECENTRALIZED PARADIGMS

This decentralization paradigm is not directly associated with operating in a coordinated (or aggregated) or individual basis. Actually, it aims to take better advantage of DERs and their most important features through the establishment of decentralized energy markets. Thus, this type of market is oriented on the peer-to-peer (P2P) approach. Here, users or nodes can adopt the role of buyers or sellers according to their situation in an energy market, in order to trade surplus electricity between them directly through a multi-layer structure (i.e. bidirectional energy flows, information and economic transactions in separate operational layers working at the same time).

On the other hand, it is important to highlight that this decentralization paradigm can also work in an aggregated or coordinated way. In fact, these synergies have created hybrid approaches such as the presented in [137] (i.e., community-based P2P and hybrid P2P energy markets), in order to leverage the operational (compatibility with existing infrastructure, scalability), economic and social benefits that these systems bring. Some studies such as those presented in [30], [138]–[140], provide evidence of these relationships.

Hence, for this paradigm, blockchain technology has had the greatest influence among those presented. This is especially due to the existence of greater compatibility according to the required structure and blockchain operation (i.e., peerto-peer). Like in the coordinated paradigm, the areas and developments in this case have been diverse: energy trading [141]–[143], energy crowdsourcing [144], demand side management [145]–[147], among others.

a: SUMMARY

In this section, we have presented a DER Integration and management paradigms where blockchain can contribute meaningfully. Here it can be evidenced that the approaches where the blockchain technology can have the most significant and diverse impact, coincide directly with the aggregation approaches paradigm (i.e. where a common benefit is prioritized, whether centralized or decentralized), where ICT plays a fundamental role since it is the basis for the features that allow the development, evolution and efficient operation of these infrastructures, in particular, the advanced metering equipment required for its expansion.

IV. BLOCKCHAIN APPLICATION IN DISTRIBUTED ENERGY RESOURCES INTEGRATION AND MANAGEMENT

In this section, we will emphasize the blockchain technology contributions for DER integration and management by presenting several recent applications and perspectives.

A. ADVANCED METERING INFRASTRUCTURE

With the Advanced Metering Infrastructure (AMI), the utility companies, consumers, prosumers, and producers can interact through the two-way communication supported by smart meters. In comparison with traditional energy meters, they can collect energy information from their owners. All this data is used for billing, control, troubleshooting, and monitoring purposes. Moreover, all this data is stored in centralized storage systems or clouds. The use of these centralized points involves potential risks of information tampering and privacy issues (i.e., identities, contact info, energy use, consumption patterns, among others), in addition to scalability and response time issues. Thus, generators and consumers still find it challenging to rely on these infrastructures. In this subsection, we summarize some of the most critical blockchain studies to address these AMI issues.

The works in [148]–[152] develop feasible applications and studies related with AMI for DER management and integration. Specifically, the authors in [148] propose a flexible, efficient, secure, and trustworthy blockchain-based access control scheme with Ethereum smart contracts (as a broker) in order to manage permissions in a distributed and reliable way. The architecture was designed in Ropsten, an official Ethereum test network, where the results confirmed that this mechanism increases the security, flexibility, efficiency, and at low cost, developing a promising solution to mitigate cyber-attacks in smart grids. This provides an interesting contribution, considering that DER are connected to these systems.

In [149] a model is introduced to explore the blockchain with smart contracts for smart grids resiliency and security. The contracts act as a broker between consumers and prosumers in order to reduce procedure costs. Once the smart meters (AMI) are connected to the blockchain system, they send a record to create a new block with a timestamp for later validation. The consumer will be charged based on the information registered in the ledger system. However, the lack of discussion in technical terms of the subject has limited the progress of this work.

Plaza et al [150], has presented a system where a smart metering infrastructure is leveraged as a secure entity for energy data trading and community governance. In addition, a blockchain-based approach is proposed for energy exchange between local energy community (LEC) members with PV power generation.

On the other hand, in [151], a model of demand-side management is introduced to decentralize smart grids. The blockchain is used to create a secure, automated and decentralized energy grid where all these nodes work independently, without the need for centralized supervision or system operator control. Additionally, it is used to store energy consumption information obtained from the smart meters integrated into the system. On the other hand, intelligent contracts offer control to validate agreements, calculate incentives or fines, and apply the rules associated with the supply and demand balance.

Meanwhile, [152] introduces a secure and reliable energy scheduling model called PPES (Privacy-Preserving Energy Scheduling) for energy service companies (ESCOs). Through blockchain and smart contracts, they address the growing privacy concerns of centralized ESCOs, which can encompass financial and behavioral information that would cause privacy issues (data security from AMI and trading) for the distributed energy market. They used a distributed optimization method to decompose the model into several individual problems to be solved with consensus algorithms and smart contracts, and probed in several case studies based on multiple energy buses.

In addition to the presented studies and projects, blockchain technology has impacted on other AMI issues, being crucial in DER management and integration tasks. Some of these fields are: IoT and blockchain deployment, privacy and security issues, AMI as an energy trading interface, smart metering and management, and artificial intelligence (AI) as well as smart city applications.

B. ELECTRIC VEHICLES

Electric vehicles (EVs) are considered as one of the main pillars of the recent energy sector development [153]. They can act as energy storage devices as well as trade energy with the power grid, charging stations and other neighboring EVs on a P2P basis [36]. Nonetheless, frequent two-way power and data communications within three possible interaction scenarios (i.e., Vehicle to Vehicle, Vehicle to Grid, and Grid to Vehicle), the short communications range, and mobility can introduce some security and privacy issues [154]. Thus, it is crucial to develop decentralized and transparent EV management and integration mechanisms. To develop such mechanisms, some of the most important contributions of intellectual production are as follows.

In this area, references [155]–[159] present possible studies related to electric vehicles fitting to blockchain environments. In [155], the authors focused on the smart contracts development empowered in permissioned blockchain systems in order to implement an EV charging framework with RES on smart grids. To deploy these contracts, an innovative energy distribution algorithm was used. Such contracts are made between energy aggregators and EVs so that they are allowed to choose the energy consumption source according to each user preference while maximizing profits. On the other hand, the energy distribution method was introduced to locate the limited energies of RES (solar PV) in EVs. However, the consensus method employed does not yet clarify the responsibility for validating transactions or incentives.

In [156], the work proposed how to store and validate EV charging-related data in blockchain and how to process EV charging payment transactions in a blockchain-based information system. Within the ongoing research activity, they evaluated some blockchain technologies regarding their applicability for this scenario. Further, they implemented a decentralized app (dApp) design and its smart contracts as a proof-of-concept for the technical feasibility of the solution, considering factors such as interoperability, data storage, trust and scalability, showing good results on simulation scenarios.

Moreover, [157] introduces a method of conserving privacy in a transparent and self-contained manner that allows EVs to find the cheapest and most viable charging stations based on energy prices and distances. This method uses blockchain when a request for supply is made for a specific energy level, where the identity of the vehicle (geographic location) is preserved and verification of the requests is increased, in order to ensuring transparency. In addition, the charging stations as an integration and management entity are also connected to the blockchain where they keep their offer records according to the requests. This method does not yet discuss the scalability of the proposal or the payment method through blockchain.

Furthermore, the authors in [158] presented a blockchainbased model for EV charging. This model incorporates smart contracts to implement the registration, scheduling, authentication and charging phases of a vehicle at a charging station. A security analysis is presented where the results show that the model meets the established objectives.

Another development under this trend is the one presented in [160], where the focus is on the infrastructure needed to charge electric vehicles, specifically the private charging piles and their corresponding operational issues. A blockchainbased framework is proposed to facilitate secure energy sharing services with cryptocurrency from a technical perspective based on the consensus mechanism performance. All operative scenarios were evaluated through game-theory approaches (joint coalition-matching game), where it was determined that this approach improves the efficiency of these charging piles. A similar approach is presented in the development presented in [161], where a novel energy blockchain system with license was realized to implement secure energy supply services for electric vehicles and charging stations, including the consensus mechanism and incentive scheme for their operation (i.e., charging and discharging tasks). The results evidenced that the proposed scheme can efficiently allocate produced energy resources to various zones with different electric loads through the movement of electric vehicles.

In addition to the presented studies and projects, blockchain technology has generated great impact on other areas of EV development, being crucial in DER management and integration. Some of these fields are: Smart charging and EV battery management, privacy and security issues, EV in VPP and crypto-currency paradigms, EV as energy trading asset, and AI as well as smart city applications.

C. ENERGY TRADING AND MARKETS

The bi-directional energy and data flows feature allow consumers to act as producers and vice versa depending on the situation (i.e., prices, energy surplus, balancing supply and demand). DER integration and management are expected to organize an increasing number of consumers, producers, and prosumers (producer + consumer roles) with different types of DER into distributed energy trading scenarios in order to get some benefits as load peak shaving, power loss decrease, lower electricity prices, amongst others. Therefore, it is necessary to perform energy trading integration along with its necessary formalities that can offer supply management, negotiation and the execution of contracts between the participants of a blockchain system. Thus, consumers and producers will be able to trade energy among themselves directly [36]. However, in traditional methods, the participants can only participate in these commercial formalities with each other indirectly through numerous third parties who experience some issues and challenges (e.g., introduction of operational costs transferred to end-users, appearance of non-competitive markets, lack of transparency and equity). In order to design a more decentralized open market and trading, the blockchain and smart contracts technology are used as discussed below.

The blockchain use for energy trading and market applications was first reported in 2014, where authors in [111], [112] introduced a virtual currency to appraise DER according to real-time metering information on production and consumption. However, existing literature does not present a comprehensive approach but focuses on specific features of blockchain-based energy markets, such as cryptocurrencies [162], [163], privacy and security [34], funding [164], optimization models [32], among others divisions. Some of the most important contributions of intellectual production are presented next.

Authors in [165] proposed a blockchain-based energy trading for EVs as an energy asset in smart parking lots. The platform architecture consists of two layers (physical and digital) that represent all the components of the system (charging stations, EVs, transformers, electric feeders, IoT devices). These components allow and support the operation of selling/buying energy among EVs (taking advantage of their dual roles of storage and consumption) in order to facilitate auditability and traceability of energy transactions. The proposed platform was implemented in Hyperledger Fabric, JavaScript and configured in Ubuntu OS. The platform provides profits for all participants and also balances energy using the electricity tariff provided by the DSO and local loads.

In [166], the authors provided a clear landscape, as well as details and technical procedures for the implementation of a pilot platform for blockchain-based P2P energy trading, considering each system node would be integrated with DER. They developed a platform connected to mobile apps as well as to different computer equipment (computers, smart meters), to execute smart contracts in Ethereum-based (tobalaba) networks, using tools such as NodeRed, Ionic, among other open-source technologies.

Under the same decentralization paradigm, some authors [137] have proposed designs for markets supporting multiple transactions between peers. In particular, the design of a full peer-to-peer market represents a challenge given the diversity and quantity of energy and services. The authors in [167] have developed a full design for a peer-to-peer market based on a multi-bilateral economic dispatch. A general formulation is presented in Eq. 1, based on [137]. This formulation expresses generation dispatch considering bilateral

transactions between peers. It is important to highlight that blockchain as an enabling technology does not include any special conditions in formulation, but it is highly correlated to the P2P paradigm.

$$\min_{D} \sum_{n \in \Omega} C_{n} \cdot \left(\sum_{m \in \omega_{n}} P_{nm} \right)$$
s.t. $P_{n} \leq \sum_{m \in \omega_{n}} P_{mn} \leq P_{n} \quad \forall n \in \Omega$

$$P_{nm} + P_{mn} = 0 \quad \forall (n, m) \in (\Omega, \omega_{n})$$

$$P_{mn} \geq 0 \quad \forall (n, m) \in (\Omega_{p}, \omega_{n})$$

$$P_{mn} \geq 0 \quad \forall (n, m) \in (\Omega_{c}, \omega_{n})$$
(1)

where, the function C_n corresponds to the production cost and P_{nm} corresponds to the transaction either sale or purchase between peers *n* and *m*. The set ω_n corresponds to all transactions done between peers. On the other hand, the sets Ω_c and Ω_p represent the consumer and producer nodes respectively. The power transactions P_{nm} and P_{mn} are equal but with opposite signs. The mathematical techniques to solve this problem includes decentralized or distributed optimization techniques such as developed in [168] and [169]. Therefore, the blockchain integrated into markets based on transactions between peers or agents represents a potential for full deployment of DER integration and management related tasks (e.g., planning, scheduling, among others).

On the other hand, inspired by energy transition trends, in [170] a breakthrough P2P energy trading scheme was proposed under the Virtual Power Plant approach using smart contracts on the blockchain Ethereum platform. P2P energy trading is a recent trend with several test projects among the power and energy societies; it facilitates the RES generation and trading in a distributed way within the local community. This scheme is based on public blockchain systems and the auctions are operated by smart contracts addressing both safety and cost issues. The implementation and execution of the smart contract in a VPP framework, including the bidding, withdrawal and development of control modules are the main features of this work.

D. ENERGY CROWDFUNDING

Since the scaling up of small and medium sized renewables, there have been big issues in reducing the capital cost of investment that end-users must bear to access DER-related devices. This is because the capital expenditures of renewable technologies are higher than conventional generation assets, even if there are no associated fuel costs. This high capital cost also carries the penalty of higher financing costs, which makes the cost of capital more critical to the economic competitiveness of renewables and their widespread adoption. On the other hand, the number of brokers and requirements for accessing possible funding sources to acquire these devices, make it difficult for many end-users to become interested in acquiring DERs.

These situations has led to the development of different studies and initiatives based on blockchain, using concepts such as decentralized finance (DeFi) to facilitate DER access and management to end-users collectively through methods such as crowdfunding. Below are some relevant studies related to this section.

References [164], [171]–[173] developed feasible applications and studies related with crowdfunding for DER integration and management. Authors in [171] propose and evaluate a blockchain-enabled crowdfunding financing option for investments in residential PV assets to further decrease LCOE values, in order to make solar PV more competitive in terms of economic viability. the findings of this study evidence that blockchain, have a positive impact in terms of reducing the financial costs and also the LCOE values of residential solar projects, which may represent a greater adoption of this resource on a small scale.

In [172], the levelized cost of residential-scale lithium-ion battery storage using blockchain-based crowdfunding is evaluated and compared to conventional financing. As a result, a tool is developed that allows multiple smaller stakeholders to offer loans with longer terms and lower interest rates, which will enable partial digital ownership of such investments and provide additional incentives for the rapid deployment and effective management of energy storage solutions.

On the other hand, entrepreneurial efforts such as those presented in [164], [173]–[175] have developed disruptive financing models that use a tool for crowd-funding small and medium-scale solar PV energy projects, eliminating the layers of intermediaries between financing and energy consumption. This tool is especially geared towards individuals who wish to use their resources to promote community development and renewable energy projects.

E. DECENTRALIZED ENERGY MANAGEMENT

The massive incursion of DER has a significantly disruptive and transforming impact on the centralized power system with unidirectional power flow that is currently in place. It is widely accepted by academia and the energy sector that a shift to a decentralized power system with bi-directional power flow is necessary for the integration of these resources [30], [54].

Due to this transformation scenario, it has become important to develop tools, frameworks, and methodologies for the proper management of DERs given their uncertainty, variability, and operational characteristics, in addition to the transitional environment facing the sector mentioned above. This management aims to facilitate a rich set of transactional energy activities between residential users with renewable energy, energy storage and flexible loads. It is also intended to guarantee reliability and efficiency, as is the case in the centralized energy approach.

Therefore, this section shows some developments in DER energy management, especially with a decentralized approach. It is important to mention that a large number of contributions have addressed how to manage DER for various objectives (e.g., energy trading, security, aggregation).

Authors in [119] developed a reliable decentralized management system for DERs using blockchain technology and smart contracts in order to integrate them to aggregators. For this purpose, several DERs were modeled and formulated a cost minimization problem for them in order to optimize their energy trading, scheduling and demand response.

Moreover, [176] proposed an integrated energy management and aggregation platform based on blockchain and smart contracts that optimizes energy flows (i.e. OPF) in a microgrid with different DERs while implementing a bilateral trading mechanism. The results evidence an increase in power grid independence by reducing imported power.

Overall, many of the developments in this field are oriented towards coordinated approaches (i.e. to proper management of DERs for the joint benefit of the end-users and the aggregator or DSO), in particular blockchain-based virtual power plants (VPPs) and smart grids (SGs) [177]. Therefore, some VPP-based developments such as those presented in [30], [178]–[182] and smart grid-based developments as [151], [183]–[185] show a high degree of adoption of both approaches in the proper management of DER from different levels (operational, financial, security, among other fields), due to these approaches are aimed at contributing to the process of decentralization of the electricity grids through interaction with the existing infrastructure, always prioritizing the economic and social benefits represented by DERs.

b: SUMMARY

In this section, we have presented a number of recent blockchain pilots, developments and entrepreneurial efforts for DER integration and management. This section has been divided into five subsections according to the different purposes that blockchain has been used for: Energy management, Electric vehicles, Advanced metering, Energy crowdfunding, and Energy markets and Trading, where it has become evident that efforts have been directed to this last area from different and unique perspectives, in order to provide an economically and technically efficient management of these resources. However, the main drawback in all these blockchain-based developments regardless of its purpose is scalability, since many of them are based on consensus mechanisms that present well-known transaction throughput issues such as PoW-based consensus mechanisms.

Besides the areas presented in this section, a comparison summary of some of the most important blockchain-based projects, pilots and applications related to DER management and integration are presented in Table 3. Considering the positive impact of business initiatives based on enabling technologies and approaches such as IoT through its concrete approach called IoE. and VPP, they have been included as divisions in this table.

V. OPPORTUNITIES AND CHALLENGES OF BLOCKCHAIN IN DER INTEGRATION AND MANAGEMENT

Blockchain presents many promising opportunities and challenges for the DER integration and management. All

blockchain technology (i.e., smart contracts, consensus mechanism, among others) has made possible for different assets to communicate without the support or regulation of a broker or intermediary.

Nonetheless, blockchain-based systems depend heavily on predetermined rules during operation time, even though no such intermediary exists, so it is necessary to ensure that this system is reliable, secure, and accurate. On the other hand, blockchain technology is still in a growth phase, where it faces several disadvantages, threats, risk and limitations such as scalability, adoption of this technology by different actors, transactions speed, among others related to its implementation and some electricity sector features in particular. In this section, we present some of the challenges and the opportunities that this technology offers for these tasks.

A. CENTRALIZATION

Currently, blockchain applications for the energy sector are in general still in the gestation stage, so it is still prone to attack from energy conglomerates that could exploit it for financial advantages [228]. Another reason for the existence of centralization is due to the architecture in the energy sector that is not yet adapted to the decentralized structure that a blockchain-based system requires [229]. This situation implies that a large investment is needed in multiple fields (e.g., technology, data flow, financial, among others) to migrate to an environment that allows interacting with this technology properly and avoid risky situations with reliable production, energy management, bad practices related to transactions, among others [40], [230].

Therefore, strict supervision must be applied under government laws as described in V-C, especially during the early stages, to ensure security and transparency in all areas. On the other hand, this situation also represents a great opportunity to create decentralized platforms additional to the financial infrastructure, where all the peers participating in the network share their computing capacity and thus avoid centralization and its issues. This development will allow an efficient operation and control of mechanisms that integrate and manage DER in power systems as well as providing other possibilities for the management of these type of systems [231].

B. TECHNICAL COMPLEXITY

Currently, the implementation of blockchain-based designs presents great challenges due to different technical issues. The processes carried out with blockchain technology involve excessive energy and computational consumption to perform the validation and security processes required for transactions in these architectures [39], [232].

This situation can lead to scalability issues, as well as being against the energy sustainability principles that they promote. On the other hand, current early-stage blockchain designs are extremely difficult to develop, deploy and maintain. For the novice user (like most potential users interested in incorporating and managing DER), performing a blockchain transaction can become an obstacle, as it requires technical knowledge

TABLE 3. Comparative summary of blockchain start-ups, applications and trials in DER integration and management in different areas: 1. Energian start-ups, applications and trials in DER integration and management in different areas: 1. Energian start-ups, applications and trials in DER integration and management in different areas: 1. Energian start-ups, applications are start-ups, applications and trials in DER integration and management in different areas: 1. Energian start-ups, applications are sta	gy
Crowdfunding, 2. Electric vehicles and storage, 3. Decentralized energy management, 4. Energy trading and markets, 5. IoT, 6. VPP.	

Blockchain App	References	Description	Area						
or project			1	2	3	4	5	6	
ChargCoin	[153], [186]	Multi-layered digital ecosystem focused on EVs and charging stations using internet of energy (IoE), in order to manage and support energy and crypto-		~		~	~		
CUBi	[187], [188]	Blockchain-based energy management platform that includes IoT (smart meter for DER integration) and AI in order to indicate enhancement opportunities.			~		~		
Dexentralize	[189]	Energy trading platform focused on self-consumption as well as massive micro-				~			
Efforce	[190]–[192]	Blockchain-based energy management and saving trading platform in order to			\checkmark	\checkmark			
		bring together clients and investors for the development of energy efficiency projects in buildings and remuneration in tokens.							
ElectronConnect (Electron)	[193]–[195]	Platform focused on decentralized markets development where different agents and energy assets trade across P2P.			~	~			
Enexis	[54], [196]	Prototyping IOTA-based cryptocurrency transactions for EV charging (currently with IBM consulting)		~					
Energy Web Chain (EWC)	[197], [198]	Public and enterprise-grade blockchain-based platform designed for enabling any energy asset owned by any customer to participate in any energy market. Also includes a Cryptoasset (energy web token)				√			
Enosi	[199]–[201]	Blockchain platform focused on smart metering and tracing for energy manage- ment and energy community services as well as decentralized markets			~	~			
ETHome	[61], [202]	Open-source blockchain-based community energy proof of concept focused in the local power sharing control and interactions, especially with ESS			~	~	~		
FlexiDAO	[203], [204]	Blockchain-based platform specialized in the management and integration of DER assets at any scale as well as decision making, currently implemented in			~				
FOHAT	[187], [205]	Iberdirola. Brazilian blockchain-based platform which integrates DERs and manages over- the sector D2D assess the large and without assessments D2D assessments.			~	√	~	~	
Greeneum	[110], [206]	Decentralized and multi-layer platform that use blockchain with machine learn-			~	~			
Grid+	[96], [109],	Decentralized peer to peer energy trading platform focused on energy cutting			~	√			
Lition	[110]	and management. Proprietary ethereum-based blockchain solution (in partnership with SAP) fo-							
Liuon	[207], [200]	cused on P2P energy transactions between consumers and producers directly, avoiding all third-parties.			v				
LO3 Energy	[104]–[106]	Blockchain-based platform that operates the Brooklyn Microgrid, letting people		~	~		~	\checkmark	
		to generate, store, buy and sell clean energy at local level. Since 2019, LOS has been working with Kyocera to develop blockchain-based VPPs to enhance its services							
ME SOLShare	[209]–[211]	Blockchain-based power sharing platform focused in "swarm electrification"			~	~	~		
MyBit	[195], [212]	Blockchain-based crowdfunding platform that distributes the small-scale PV	~						
NRGCoin	[111]–[113]	Decentralized virtual currency and multi-layered digital ecosystem pilot focused				~			
PowerLedger	[82], [84],	Multi-layer blockchain-based platform focused on new energy markets enabling			↓	 √		 ✓ 	
Power2Peer	[85]	Multi-layer and blockchain-based platform for P2P energy trading using local			\checkmark	√	√		
Pulon Natwork	[01] [02]	PV in order to bring local resilience to the grid.							
F yion Network	[91]-[93]	play the role of ICT and trading.		v	v				
(MotionWerk)	[114]-[116]	Blockchain-based platform for electric vehicle charging/discharging manage- ment and coordination for a better experience in Europe		√			√		
Solara	[214], [215]	Blockchain-based platform for secure solar energy data management in order to enable new financial models.			~		~		
SolarChange (So- larCoin)	[216]–[218]	Blockchain-based app suite that includes cryptocurrencies as an incentive for solar energy production and crowdfunding, as well as energy management features	√		√				
Slock.it	[101]–[103]	Blockchain and IoT based platform focused on peer-to-peer sharing of IoT DER assets in cooperation with Energy Web Foundation.				~	v		
Synergy (Electrify Acia)	[219]–[221]	P2P energy trading platform using blockchain, connecting energy producers (or				\checkmark			
Sun Exchange	[97]–[100]	Blockchain-based crowdfunding platform to fund PV installation which supply	 ✓ 						
		energy to public benefit buildings in South Africa (e.g hospitals, schools, among others).							
Transactive Energy Initiative	[222]–[224]	Blockchain-based platform of DER transactive energy under pilot development focused on P2P trading and markets for energy communities in Medellín, daveloped by ElA togetber with UCL and ERCO.			✓ ¯				
Unergy.io	[175]	Blockchain-based collaborative economy platform that enables secure crowd-	\checkmark			-			
Verv VLUX	[201], [225]	funding to invest in or finance solar energy projects at any scale. Developing platform for p2p (private blokchain) energy trading at grid edge. Its				↓	↓		
WePower	[226] [227]	focus is on local energy communities.							
, mer ower	[[220], [227]	through energy tokenization.	`		*				

and several sophisticated steps. This added to the standardization problem mentioned in section V-E, making the ICT systems too complex for the end-users who then avoid using them.

Therefore, it is necessary that the development of future blockchain-based applications adopt plug-and-play infrastructures that are more simpler and user-friendly to make a transaction (financial, data flows, etc.) more common among end-users. Likewise, in relation to the high computational and energy consumption of these systems, there is a long way to go in the development of specific and lighter consensus mechanisms for the unique dynamics of DER management aimed at lower energy consumption, while maintaining the privacy, security and decentralization standards required for these systems, and developing compatibility with enabling technologies such as IoT [233], VPP, among others.

On the other hand, it is crucial to move towards a more interconnected and technology-enabled smart grid. Currently, many companies in the energy sector are forced to manufacture their own smart meters, making it a gradual process that requires a sustained effort to install all the necessary devices (meters, poles, cables, among others) to facilitate inter-connectivity.

C. POLICY SUPPORT

Energy applications for blockchain-based DER integration and management in many countries still suffer from several gray areas, which result in institutional [38], [234] and socioeconomic [235] consequences. There is a lack of information on potential distributed energy consumer markets as well as implementing restrictive energy markets, which would be valuable for the formation of an appropriate institutional and regulatory framework.

This situation has prompted regulators to support the active participation of users in energy markets, the creation of institutions, and well-defined laws that support DER integration and management. However, when it comes to substantial changes in the main power grid and the core business, these same entities do not yet support key issues for DER integration, such as direct energy trading and management between consumers or prosumers.

Therefore, precise legal definitions as well as well-defined regulatory and institutional frameworks are needed to mitigate the level of uncertainty associated with blockchain in this task. Thus, new types of agreements, market mechanisms, institutions, and policies must be introduced, in order to support DER integration and management including all stakeholders [236], which are currently heavily restricted due to the high risks that this technology can still represent [237]. In this way, although blockchain technology has shown great value in managing and integrating DER into the network (i.e., microgrids, electric vehicle management, among others) there is still great resistance to the adoption at regulatory level of this type of technology on a large scale.

D. DEVELOPMENT COSTS

Blockchain technology development for DER integration and management requires high costs in order to restructure the current grid, upgrading the meters for the transactions with smart contracts. Investing in information and communication technologies is required for advanced metering interfaces in order to interact with these resources of conventional power grids in a hybrid way. All these costs keep the grid operators from investing in the blockchain deployment, opting instead for alternatives with a higher maturity degree and lower costs (e.g., telemetry).

However, the blockchain technology has passed several proof-of-concept phases that show its feasibility. For this reason, it must compete with more mature proposals in relation to its profitability and scalability. There is the possibility of reducing the cost of storage (one of the highest deployment costs for this technology) by storing the actual data in side chains (as a subsidiary blockchain) and operating the main blockchain as a control layer instead of a storage layer [41]. It is important to mention that it is necessary to focus on a lighter overall development of these systems (e.g., aggregators [30], IoT-based systems, among others), in order to reduce both transaction costs and physical device specifications needed to carry them out.

E. STANDARDIZATION

A series of mechanisms, protocols and technological solutions are being developed for the power grid based on blockchain. However, the main challenge faced by these systems for integrating and managing DER is that it lacks widely-accepted standards due to the lack of maturity of this technology.

Ultimately, this situation prevents the integration of different devices as smart meters, IoT devices, electric cyberphysical systems, electric vehicles, and other entities related to the power grid [238], [239] as well as new issues related to utility operators and electricity markets [240]. However, this issue also represent a great opportunity for the development of standards/rules related to success cases that reflect good results (e.g., demand supply balance, forecasting grid requirements, self-adjusting power consumption, among others) and thus establish a road map for new developments in this area.

Thus, this development should be oriented towards creation of a safe, open and easily applicable framework, which together with other technologies and standards of quality (e.g. IEEE, IEC, among others) [241], [242] will give rise to holistic and applicable guidelines related to the DER integration and management considering all features that this process entails.

c: SUMMARY

In this section the main challenges and opportunities related to the management and integration of DER through blockchain technology have been presented. It can be seen that great efforts must still be made in multiple fields in

a coordinated way, since few properly regulated sandboxes or pilots have been found that allow testing these developments in order to address these issues. Therefore, public policy in different countries should promote an expansion in blockchain-based developments in a fair way, with a joint objective and duly regulated along with technical development (i.e. to make the centralization of the energy sector more flexible from the regulatory and technological perspectives), where alternatives can be generated to reduce development costs, as well as the known problems of standardization and scalability according to the context of each place, so that these developments become viable both for business initiatives and new business models and for the communities that appropriate the DER from a decentralized approach.

VI. DISCUSSION

An explanation of blockchain technology applied to DER integration and management is carried out in the previous sections. Based on the insights provided in Sections II, IV and V, it is evident that the blockchain contribution to the future development of DER integration and management could be very significant.

This is evidenced by the exponential increase and inclusion in recent years of different DER in small-scale distribution grids, as well as the growing number of projects and start-ups in the energy sector that are involved in projects with this technology as shown in Table 3. On the other hand, based on its own nature, blockchain technology in conjunction with other enabling technologies such as AI, IoT, VPP, among others, can improve the performance, efficiency and stability of the grid, considering DER variability, uncertainty and decentralization.

However, it is important to highlight that the transition cost to decentralization with special focus on power systems operation and markets, and corresponding blockchain implementation can be very high (due to the multi-layered nature of this process), but the benefits are numerous and will ultimately reduce electricity costs. By migrating to a decentralized, blockchain-based system, the following benefits could be available to the current electricity sector:

- Advanced support for DER integration and local energy management (metering and efficiency services).
- Flexible, sustainable, and secure energy trading without brokers.
- Real-time statistics on energy consumption patterns, generation prices, time-of-use pricing and suggestions on improving energy efficiency.
- Peak load shaving based on flexible and convenient hours of operation, as well as rewards (e.g., crypto-assets, energy credits or tokens) both locally and on a large scale.
- Extended support to ancillary energy infrastructures and services (e.g., city surveillance systems, street lighting, DR programs, EV charging stations, among others).

In addition to the aforementioned benefits, with the advancement in the energy sector digitization, DER management and integration processes have also improved with respect to reliability, availability, resilience, stability, security, and privacy [243]–[245]. The following paragraphs analyze the blockchain contribution in the enhancement of these features.

Availability, reliability and stability are the most critical features of the management of the grid and its resources (whether distributed or centralized). This is due to their increasing complexity in the modern energy sector, meeting the objectives of these features becomes more difficult. In this sense, blockchain technology in conjunction with tools such as smart grids, could provide a more complete control and monitoring of DER in power grids in order to solve the problems related to aggravated grid congestion, and DER uncertainty and variability [14], [246], [247].

The issues associated with security, privacy and resilience, derive from the P2P energy trading on the grid as an alternative for managing and integrating DER, even under unexpected disruptive events. The trust and transparency features of blockchain ensure that the data flow (i.e., financial and data flows) between peers is always secure and occurs through specific channels based on the agreement defined in mechanisms such as smart contracts contributing to an orderly management of available resources in order to provide data integrity and confidentiality.

Overall, the blockchain research revealed a greater scope in DER integration and management processes when it has been accompanied with other paradigms linked to coordinated and community-based aggregation schemes such as IoT, VPP, among others, in order to improve its features and effectiveness. Likewise, the fundamental role of ICT in the development of blockchain is evident, since it supports the operability, evolution, and future path of its development, where scalability, standardization of mechanisms, support at the policy level, and development costs are the main issues, where a joint effort at the level of regulation and technological progress is needed.

VII. CONCLUSION

Blockchain technology in DER management and integration is a new and emerging area that has attracted attention due to its effects on power generation resilience. In this paper, a comprehensive review of the current blockchain landscape for DER management and integration is presented. First, the theoretical background of blockchain technology is presented. Next, some recent blockchain success cases (pilots, trials, and start-ups) in DER management and integration tasks are summarized. Finally, the opportunities and challenges of DER integration as well as blockchain features in order to understand the reasons for using it in these tasks are examined. This development together with other technologies has great applicability in DER mainstreaming and therefore in the energy transition process that the world needs to carry out. According to the literature review performed, the following conclusions are presented in order to contribute to the blockchain development for the management and integration of DER:

- Blockchain technology is very promising for DER management and integration considering the different applications geared to this task. However, it is noted that blockchain technology is still at an early stage of development. This implies that deep changes are needed in a comprehensive way in relation to the infrastructure and technology of centralized power grids. Currently, there is no clear and standardized scheme within regulatory market and institutional frameworks to guide its deployment.
- Blockchain technology relies on the joint use of other technologies (e.g. IoT, AMI, AI, among others) and approaches such as VPP, smart grids, local energy communities, amongst others, to take full advantage of the utilities it can offer. However, there is no unified approach on how to implement these technologies in a coordinated fashion; currently this is occurring based on the environment of the DER-based application or system where it is deployed.
- One of the key research objectives for developing blockchain-based applications for DER management and integration is to incorporate flexibility in multiple fields (economic, operational and market) within the energy sector. Therefore, the future direction of research should focus on multidisciplinary efforts (regulatory, scientific, economic, market, among others) so that its contribution may be comprehensive, which may represent better results in the short to medium term.

REFERENCES

- M. U. Gurmani, T. Sultana, A. Ghaffar, M. Azeem, Z. Abubaker, H. Farooq, and N. Javaid, "Energy trading between prosumer and consumer in P2P network using blockchain," in *Proc. Int. Conf. P2P, Parallel, Grid, Cloud Internet Comput.* Cham, Switzerland: Springer, 2019, pp. 875–886.
- [2] J. Newman and P. MacDougall, "Increasing DER integration through discrete intraday settlements," *Electr. J.*, vol. 34, no. 4, May 2021, Art. no. 106932, doi: 10.1016/j.tej.2021.106932.
- [3] J. D. Prado, W. Qiao, L. Qu, and J. Agüero, "The next-generation retail electricity market in the context of distributed energy resources: Vision and integrating framework," *Energies*, vol. 12, no. 3, p. 491, Feb. 2019, doi: 10.3390/en12030491.
- [4] Z. Li, S. Bahramirad, A. Paaso, M. Yan, and M. Shahidehpour, "Blockchain for decentralized transactive energy management system in networked microgrids," *Electr. J.*, vol. 32, no. 4, pp. 58–72, May 2019, doi: 10.1016/j.tej.2019.03.008.
- [5] T. S. Babu, K. R. Vasudevan, V. K. Ramachandaramurthy, S. B. Sani, S. Chemud, and R. M. Lajim, "A comprehensive review of hybrid energy storage systems: Converter topologies, control strategies and future prospects," *IEEE Access*, vol. 8, pp. 148702–148721, 2020, doi: 10.1109/ACCESS.2020.3015919.
- [6] E. A. Soto, L. B. Bosman, E. Wollega, and W. D. Leon-Salas, "Peer-topeer energy trading: A review of the literature," *Appl. Energy*, vol. 283, Feb. 2021, Art. no. 116268.
- [7] N. D. Hatziargyriou and G. E. Asimakopoulou, "DER integration through a monopoly DER aggregator," *Energy Policy*, vol. 137, Feb. 2020, Art. no. 111124, doi: 10.1016/j.enpol.2019.111124.

- [8] Y. Wu, Y. Wu, H. Cimen, J. C. Vasquez, and J. M. Guerrero, "P2P energy trading: Blockchain-enabled P2P energy society with multiscale flexibility services," *Energy Rep.*, vol. 8, pp. 3614–3628, Nov. 2022. [Online]. Available: https://www.sciencedirect.com/ science/article/pii/S2352484722003213
- [9] H. Wang, S. Riaz, and P. Mancarella, "Integrated techno-economic modeling, flexibility analysis, and business case assessment of an urban virtual power plant with multi-market co-optimization," *Appl. Energy*, vol. 259, Feb. 2020, Art. no. 114142, doi: 10.1016/j.apenergy.2019. 114142.
- [10] L. Blackhall, G. Kuiper, L. Nicholls, and P. Scott, "Optimising the value of distributed energy resources," *Electr. J.*, vol. 33, no. 9, Nov. 2020, Art. no. 106838, doi: 10.1016/j.tej.2020.106838.
- [11] J. C. D. Prado and W. Qiao, "A stochastic decision-making model for an electricity retailer with intermittent renewable energy and shortterm demand response," *IEEE Trans. Smart Grid*, vol. 10, no. 3, pp. 2581–2592, May 2019.
- [12] AEMO. (Apr. 2019). Technical Integration of Distributed Energy Resources: Improving DER Capabilities to Benefit Consumer and the Power System. [Online]. Available: https://www.aemo.com.au/-/media/Files/Electricity/NEM/DER/2019/Technical-Integration/Technical-Integration-of-DER-Report.pdf
- [13] J.-H. Huh and S.-K. Kim, "The blockchain consensus algorithm for viable management of new and renewable energies," *Sustainability*, vol. 11, no. 11, p. 3184, Jun. 2019, doi: 10.3390/su11113184.
- [14] N. M. Kumar, A. A. Chand, M. Malvoni, K. A. Prasad, K. A. Mamun, F. R. Islam, and S. S. Chopra, "Distributed energy resources and the application of AI, IoT, and blockchain in smart grids," *Energies*, vol. 13, no. 21, p. 5739, Nov. 2020, doi: 10.3390/en13215739.
- [15] M. Khorasany, A. Dorri, R. Razzaghi, and R. Jurdak, "Lightweight blockchain framework for location-aware peer-to-peer energy trading," *Int. J. Electr. Power Energy Syst.*, vol. 127, May 2021, Art. no. 106610, doi: 10.1016/j.ijepes.2020.106610.
- [16] E. Lannoye, "Activation of distributed energy resources in the energy market," Electr. Power Res. Inst. (EPRI), Washington, DC, USA, Tech. Rep., 2019.
- [17] M. F. Dynge, P. C. D. Granado, N. Hashemipour, and M. Korpås, "Impact of local electricity markets and peer-to-peer trading on low-voltage grid operations," *Appl. Energy*, vol. 301, Nov. 2021, Art. no. 117404, doi: 10.1016/j.apenergy.2021.117404.
- [18] M. Lotfi, C. Monteiro, M. Shafie-Khah, and J. P. Catal ao, "Transition toward blockchain-based electricity trading markets," in *Blockchain-Based Smart Grids*. Amsterdam, The Netherlands: Elsevier, 2020, pp. 43–59.
- [19] A. S. Yahaya, N. Javaid, F. A. Alzahrani, A. Rehman, I. Ullah, A. Shahid, and M. Shafiq, "Blockchain based sustainable local energy trading considering home energy management and demurrage mechanism," *Sustainability*, vol. 12, no. 8, p. 3385, Apr. 2020.
- [20] M. E. Honarmand, V. Hosseinnezhad, B. Hayes, and P. Siano, "Local energy trading in future distribution systems," *Energies*, vol. 14, no. 11, p. 3110, May 2021.
- [21] T. Woolf, C. Lane, M. Whited, C. Neme, M. Alter, S. Fine, K. Rábago, S. Schiller, K. Strickland, and B. Chew, "National standard practice manual for benefit-cost analysis of distributed energy resources," Nat. Energy Screening Project (NESP), Tech. Rep., Aug. 2020. [Online]. Available: https://www.nationalenergyscreeningproject.org/wpcontent/uploads/2020/08/NSPM-Summary_08-24-2020.pdf
- [22] D. Sanwal, D. Gusain, G. Leena, and M. Verma, "Distributed energy resources: Issues and challenges," *Int. J. Adv. Res.*, vol. 6, pp. 1127–1132, Mar. 2018.
- [23] C. Liu, X. Zhang, K. K. Chai, J. Loo, and Y. Chen, "A survey on blockchain-enabled smart grids: Advances, applications and challenges," *IET Smart Cities*, vol. 3, no. 2, pp. 56–78, Jun. 2021.
- [24] Z. Zeng, M. Dong, W. Miao, M. Zhang, and H. Tang, "A datadriven approach for blockchain-based smart grid system," *IEEE Access*, vol. 9, pp. 70061–70070, 2021, doi: 10.1109/ACCESS.2021. 3076746.
- [25] M. J. Thompson, H. Sun, and J. Jiang, "Blockchain-based peer-to-peer energy trading method," *CSEE J. Power Energy Syst.*, early access, Sep. 10, 2021, doi: 10.17775/CSEEJPES.2021.00010.
- [26] S. Küfeoğlu, "Blockchain applications and peer-to-peer tradings," in *The Home of the Future*. Cham, Switzerland: Springer, 2021, pp. 221–258.
- [27] P. W. Khan and Y.-C. Byun, "Blockchain-based peer-to-peer energy trading and charging payment system for electric vehicles," *Sustainability*, vol. 13, no. 14, p. 7962, Jul. 2021.

- [28] J. G. Song, E. S. Kang, H. W. Shin, and J. W. Jang, "A smart contract-based P2P energy trading system with dynamic pricing on ethereum blockchain," *Sensors*, vol. 21, no. 6, p. 1985, Mar. 2021, doi: 10.3390/s21061985.
- [29] J. Lu, S. Wu, H. Cheng, and Z. Xiang, "Smart contract for distributed energy trading in virtual power plants based on blockchain," *Comput. Intell.*, vol. 37, no. 3, pp. 1445–1455, Aug. 2021.
- [30] Q. Yang, H. Wang, T. Wang, S. Zhang, X. Wu, and H. Wang, "Blockchain-based decentralized energy management platform for residential distributed energy resources in a virtual power plant," *Appl. Energy*, vol. 294, Jul. 2021, Art. no. 117026.
- [31] W. Tushar, C. Yuen, T. K. Saha, T. Morstyn, A. C. Chapman, M. J. E. Alam, S. Hanif, and H. V. Poor, "Peer-to-peer energy systems for connected communities: A review of recent advances and emerging challenges," *Appl. Energy*, vol. 282, Jan. 2021, Art. no. 116131, doi: 10.1016/j.apenergy.2020.116131.
- [32] T. Cioara, M. Antal, V. T. Mihailescu, C. D. Antal, I. M. Anghel, and D. Mitrea, "Blockchain-based decentralized virtual power plants of small prosumers," *IEEE Access*, vol. 9, pp. 29490–29504, 2021, doi: 10.1109/ACCESS.2021.3059106.
- [33] C. M. S. Ferreira, R. A. R. Oliveira, J. S. Silva, and C. F. M. D. C. Cavalcanti, "Blockchain for machine to machine interaction in industry 4.0," in *Blockchain Technology for Industry 4.0*. Singapore: Springer, 2020, pp. 99–116.
- [34] A. Bischi, M. Basile, D. Poli, C. Vallati, F. Miliani, G. Caposciutti, M. Marracci, G. Dini, and U. Desideri, "Enabling low-voltage, peer-to-peer, quasi-real-time electricity markets through consortium blockchains," *Appl. Energy*, vol. 288, Apr. 2021, Art. no. 116365, doi: 10.1016/j.apenergy.2020.116365.
- [35] M. Andoni, V. Robu, D. Flynn, S. Abram, D. Geach, D. Jenkins, P. McCallum, and A. Peacock, "Blockchain technology in the energy sector: A systematic review of challenges and opportunities," *Renew. Sustain Energy Rev.*, vol. 100, pp. 143–174, Feb. 2019, doi: 10.1016/j.rser.2018.10.014.
- [36] M. B. Mollah, J. Zhao, D. Niyato, K.-Y. Lam, X. Zhang, A. M. Y. M. Ghias, L. H. Koh, and L. Yang, "Blockchain for future smart grid: A comprehensive survey," *IEEE Internet Things J.*, vol. 8, no. 1, pp. 18–43, Jan. 2021.
- [37] Q. Wang and M. Su, "Integrating blockchain technology into the energy sector—From theory of blockchain to research and application of energy blockchain," *Comput. Sci. Rev.*, vol. 37, Aug. 2020, Art. no. 100275.
- [38] A. Ahl, M. Yarime, K. Tanaka, and D. Sagawa, "Review of blockchain-based distributed energy: Implications for institutional development," *Renew. Sustain. Energy Rev.*, vol. 107, pp. 200–211, Jun. 2019.
- [39] J. Wu and N. Tran, "Application of blockchain technology in sustainable energy systems: An overview," *Sustainability*, vol. 10, no. 9, p. 3067, Aug. 2018.
- [40] A. Miglani, N. Kumar, V. Chamola, and S. Zeadally, "Blockchain for internet of energy management: Review, solutions, and challenges," *Comput. Commun.*, vol. 151, pp. 395–418, Feb. 2020.
- [41] T. Alladi, V. Chamola, J. J. P. C. Rodrigues, and S. A. Kozlov, "Blockchain in smart grids: A review on different use cases," *Sensors*, vol. 19, no. 22, p. 4862, Nov. 2019, doi: 10.3390/s19224862.
- [42] C. Yapa, C. de Alwis, M. Liyanage, and J. Ekanayake, "Survey on blockchain for future smart grids: Technical aspects, applications, integration challenges and future research," *Energy Rep.*, vol. 7, pp. 6530–6564, Nov. 2021.
- [43] J. Guerrero, D. Gebbran, S. Mhanna, A. C. Chapman, and G. Verbič, "Towards a transactive energy system for integration of distributed energy resources: Home energy management, distributed optimal power flow, and peer-to-peer energy trading," *Renew. Sustain. Energy Rev.*, vol. 132, Oct. 2020, Art. no. 110000.
- [44] J. Bao, D. He, M. Luo, and K.-K.-R. Choo, "A survey of blockchain applications in the energy sector," *IEEE Syst. J.*, vol. 15, no. 3, pp. 3370–3381, Sep. 2021.
- [45] Y. Guo, Z. Wan, and X. Cheng, "When blockchain meets smart grids: A comprehensive survey," 2021, arXiv:2109.14130.
- [46] T. Wang, H. Hua, Z. Wei, and J. Cao, "Challenges of blockchain in new generation energy systems and future outlooks," *Int. J. Electr. Power Energy Syst.*, vol. 135, Feb. 2022, Art. no. 107499.
- [47] C. Burger, A. Kuhlman, P. Richard, and J. Weinmann, "Blockchain in the energy transition a survey among decision-makers in the German energy industry," DENA German Energy Agency, Berlin, Germany, Tech. Rep., 2015, vol. 60.

- [48] M. Rauchs, A. Glidden, B. Gordon, G. C. Pieters, M. Recanatini, F. Rostand, K. Vagneur, and B. Z. Zhang, "Distributed ledger technology systems: A conceptual framework," Tech. Rep., 2018.
- [49] A. M. Alonso, E. Martin, A. Mateo, F. J. Nogales, C. Ruiz, and A. Veiga, "Clustering electricity consumers: Challenges and applications for operating smart grids," *IEEE Power Energy Mag.*, vol. 20, no. 3, pp. 54–63, May 2022.
- [50] S. Nakamoto, "Bitcoin: A peer-to-peer electronic cash system," Decentralized Bus. Rev., p. 21260, Oct. 2008.
- [51] K. Ioannis, G. Raimondo, G. Dimitrios, D. Gioia, K. Georgios, N. Ricardo, and N.-F. Igor, "Blockchain in energy communities," *Centro Comum de Investigação da Comissão Europeia*, vol. 344003, p. 62, 2017.
- [52] I. Bashir, Mastering Blockchain: Distributed Ledger Technology, Decentralization, and Smart Contracts Explained, 2nd ed. Birmingham, U.K.: Packt Publishing, 2018.
- [53] B. Marchi, S. Zanoni, I. Ferretti, L. E. Zavanella, and M. Pasetti, "The disruptive potential of blockchain technologies in the energy sector," in *Proc. ECEEE Summer Study*, 2019, pp. 899–906.
- [54] Blockchain—Innovation Landscape Brief, Int. Renew. Energy Agency, Abu Dhabi, United Arab Emirates, 2019.
- [55] J. Mattila, T. Seppälä, C. Naucler, R. Stahl, M. Tikkanen, A. Bådenlid, and J. Seppälä, "Industrial blockchain platforms: An exercise in use case development in the energy industry," ETLA, Helsinki, Finland, Tech. Rep., Nov. 2016.
- [56] A. Pieroni, N. Scarpato, L. Di Nunzio, F. Fallucchi, and M. Raso, "Smarter city: Smart energy grid based on blockchain technology," *Int. J. Adv. Sci. Eng. Inf. Technol*, vol. 8, no. 1, pp. 298–306, 2018.
- [57] D. Tapscott and A. Tapscott, "The impact of the blockchain goes beyond financial services," *Harvard Bus. Rev.*, vol. 10, no. 7, pp. 1–4, 2016.
- [58] K. Heires, "The risks and rewards of blockchain technology," *Risk Manage.*, vol. 63, no. 2, pp. 4–7, 2016.
- [59] R. Yang, R. Wakefield, S. Lyu, S. Jayasuriya, F. Han, X. Yi, X. Yang, G. Amarasinghe, and S. Chen, "Public and private blockchain in construction business process and information integration," *Autom. Construct.*, vol. 118, Oct. 2020, Art. no. 103276, doi: 10.1016/j.autcon.2020.103276.
- [60] M. Fan and X. Zhang, "Consortium blockchain based data aggregation and regulation mechanism for smart grid," *IEEE Access*, vol. 7, pp. 35929–35940, 2019, doi: 10.1109/ACCESS.2019.2905298.
- [61] K.-C. Li, X. Chen, H. Jiang, and E. Bertino, *Essentials of Blockchain Technology*. Boca Raton, FL, USA: CRC Press, Nov. 2019.
- [62] M. Aloqaily, A. Boukerche, O. Bouachir, F. Khalid, and S. Jangsher, "An energy trade framework using smart contracts: Overview and challenges," *IEEE Netw.*, vol. 34, no. 4, pp. 119–125, Jul. 2020, doi: 10.1109/MNET.011.1900573.
- [63] M. Shafie-Khah, Ed., Blockchain-Based Smart Grids, 1st ed. Amsterdam, The Netherlands: Elsevier, 2020.
- [64] K. Christidis and M. Devetsikiotis, "Blockchains and smart contracts for the Internet of Things," *IEEE Access*, vol. 4, pp. 2292–2303, 2016.
- [65] A. Cohn, T. West, and C. Parker, "Smart after all: Blockchain, smart contracts, parametric insurance, and smart energy grids," *Georgetown Law Technol. Rev.*, vol. 1, no. 2, pp. 273–304, 2017.
- [66] Z. Zheng, S. Xie, H.-N. Dai, W. Chen, X. Chen, J. Weng, and M. Imran, "An overview on smart contracts: Challenges, advances and platforms," *Future Gener. Comput. Syst.*, vol. 105, pp. 475–491, Dec. 2020, doi: 10.1016/j.future.2019.12.019.
- [67] M. L. Di Silvestre, P. Gallo, J. M. Guerrero, R. Musca, E. R. Sanseverino, G. Sciumè, J. C. Vásquez, and G. Zizzo, "Blockchain for power systems: Current trends and future applications," *Renew. Sustain. Energy Rev.*, vol. 119, Mar. 2020, Art. no. 109585, doi: 10.1016/j.rser.2019. 109585.
- [68] C. T. Nguyen, D. T. Hoang, D. N. Nguyen, D. Niyato, H. T. Nguyen, and E. Dutkiewicz, "Proof-of-stake consensus mechanisms for future blockchain networks: Fundamentals, applications and opportunities," *IEEE Access*, vol. 7, pp. 85727–85745, 2019.
- [69] A. Corusa, J. Predel, and N. Schöne, "Eine marktübersicht der blockchain in der energiewirtschaft. von der idee zum geschäftsmodell, von der technologie zur aktuellen anwendung," Technische Universität Berlin, Berlin, Germany, Tech. Rep., Aug. 2020.
- [70] M. K. Thukral, "Emergence of blockchain-technology application in peer-to-peer electrical-energy trading: A review," *Clean Energy*, vol. 5, no. 1, pp. 104–123, Mar. 2021.
- [71] S. King and S. Nadal, "PPCoin: Peer-to-peer crypto-currency with proofof-stake," Tech. Rep., Aug. 2012, vol. 19, no. 1.

- [72] D. Mingxiao, M. Xiaofeng, Z. Zhe, W. Xiangwei, and C. Qijun, "A review on consensus algorithm of blockchain," in Proc. IEEE Int. Conf. Syst., Man, Cybern. (SMC), Oct. 2017, pp. 2567-2572. [Online]. Available: http://ieeexplore.ieee.org/document/8123011/
- [73] B. Cao, Z. Zhang, D. Feng, S. Zhang, L. Zhang, M. Peng, and Y. Li, "Performance analysis and comparison of PoW, PoS and DAG based blockchains," Digit. Commun. Netw., vol. 6, no. 4, pp. 480-485, Nov. 2020
- [74] F. Yang, W. Zhou, Q. Wu, R. Long, N. N. Xiong, and M. Zhou, "Delegated proof of stake with downgrade: A secure and efficient blockchain consensus algorithm with downgrade mechanism," IEEE Access, vol. 7, pp. 118541-118555, 2019.
- [75] T. M. Fernández-Caramés and P. Fraga-Lamas, "A review on the use of blockchain for the Internet of Things," IEEE Access, vol. 6, pp. 32979-33001, 2018.
- [76] S. M. S. Saad and R. Z. R. M. Radzi, "Comparative review of the blockchain consensus algorithm between proof of stake (POS) and delegated proof of stake (DPOS)," Int. J. Innov. Comput., vol. 10, no. 2, pp. 27-32, Nov. 2020, doi: 10.11113/ijic.v10n2.272
- [77] S. D. Angelis, L. Aniello, R. Baldoni, F. Lombardi, A. Margheri, and V. Sassone, "PBFT vs proof-of-authority: Applying the CAP theorem to permissioned blockchain," Tech. Rep., 2018, p. 11. [78] Y. Yu, Y. Guo, W. Min, and F. Zeng, "Trusted transactions in micro-grid
- based on blockchain," *Energies*, vol. 12, no. 10, p. 1952, May 2019. [79] B. Lashkari and P. Musilek, "A comprehensive review of blockchain
- consensus mechanisms," IEEE Access, vol. 9, pp. 43620-43652, 2021, doi: 10.1109/ACCESS.2021.3065880.
- S. Dziembowski, S. Faust, V. Kolmogorov, and K. Pietrzak, "Proofs of [80] space," in Proc. Annu. Cryptol. Conf. Berlin, Germany: Springer, 2015,
- pp. 585–605. V. Sharma and N. Lal, "A novel comparison of consensus algorithms in [81] blockchain," Adv. Appl. Math. Sci., vol. 20, no. 1, pp. 1-13, 2020.
- [82] B. Carson, G. Romanelli, P. Walsh, and A. Zhumaev, "Blockchain beyond the hype: What is the strategic business value," McKinsey Company, Atlanta, GA, USA, Tech. Rep., 2018, pp. 1–13. C. Grimes, "Power to the people," *Electr. Connection*, pp. 28–29,
- [83] Winter 2018.
- [84] PowerLedger. (2018). Power Ledger Whitepaper. [Online]. Available: https://cdn2.hubspot.net/hubfs/4519667/Documents%20/Power %20Ledger%20Whitepaper.pdf
- [85] PowerLedger. Accessed: Jul. 23, 2021. [Online]. Available: https://www. powerledger.io/platform
- [86] A. Ekblaw and A. Lippman, "SolarCoin: Making blockchain mining open green and ready for micropayments," Version 1.0, PubPub, Tech. Rep., 2016.
- [87] J. Dispenza, C. Garcia, and R. Molecke, "Energy efficiency coin (EECoin) a blockchain asset class pegged to renewable energy markets," EnLedger Corp., Lewes, DE, USA, Tech. Rep., 2017, vol. 1. [88] J. Ferreira and A. Martins, "Building a community of users for open
- market energy," Energies, vol. 11, no. 9, p. 2330, Sep. 2018.
- [89] GridSingularity. (Apr. 2020). An Energy Exchange Engine for Local Energy Marketplaces, Medium. [Online]. Available: https:// gridsingularity.medium.com/an-energy-exchange-engine-for-localenergy-marketplaces-28d5be23705e
- [90] M. Stephant, K. Hassam-Ouari, D. Abbes, A. Labrunie, and B. Robyns, "A survey on energy management and blockchain for collective selfconsumption," in Proc. 7th Int. Conf. Syst. Control (ICSC), Oct. 2018, pp. 237–243.
- [91] Pylon Network. (Dec. 2018). Whitepaper Pylon V2.0. [Online]. Available: https://pylon-network.org/wp-content/uploads/2019/02/ WhitePaper_PYLON_v2_ENGLISH-1.pdf
- [92] A. Adeyemi, M. Yan, M. Shahidehpour, C. Botero, A. V. Guerra, N. Gurung, L. C. Zhang, and A. Paaso, "Blockchain technology applications in power distribution systems," Electr. J., vol. 33, no. 8, Oct. 2020, Art. no. 106817.
- [93] F. L. de Alba, A. González-Briones, P. Chamoso, T. Pinto, Z. Vale, and J. M. Corchado, "A P2P electricity negotiation agent systems in urban smart grids," in Distributed Computing and Artificial Intelligence, Special Sessions, 17th International Conference. Cham, Switzerland: Springer, Jul. 2020, pp. 97-106, doi: 10.1007/978-3-030-53829-3_9.
- [94] Everty. Charging Station Management for Residential, Business or Public Locations. Accessed: Jan. 15, 2021. [Online]. Available: https://everty.com.au/
- [95] Energo Labs: Decentralized Autonomous Energy System, Energo Labs, Shanghai, China, 2017.
- [96] T. Baumann, "Blockchain for planetary stewardship: Using the disruptive force of distributed ledger technology to fight climate disruption," Tech. Rep., Jan. 2018.

- [97] A. Woodhall, "How blockchain can democratize global energy supply," in Transforming Climate Finance and Green Investment With Blockchains. Amsterdam, The Netherlands: Elsevier, 2018, pp. 65-82, doi: 10.1016/b978-0-12-814447-3.00005-7
- [98] M. J. Bürer, M. de Lapparent, V. Pallotta, M. Capezzali, and M. Carpita, "Use cases for blockchain in the energy industry opportunities of emerging business models and related risks," Comput. Ind. Eng., vol. 137, Nov. 2019, Art. no. 106002, doi: 10.1016/j.cie.2019. 106002
- [99] S. Pizzi, L. Corbo, and A. Caputo, "Fintech and SMEs sustainable business models: Reflections and considerations for a circular economy," J. Cleaner Prod., vol. 281, Jan. 2021, Art. no. 125217, doi: 10.1016/j.jclepro.2020.125217.
- [100] C. A. D. Valdivia and M. Poblet, "Blockchain governance: Innovating sustainable energy financing in global south," SSRN Electron. J., pp. 1-21, Feb. 2021, doi: 10.2139/ssrn.3777348.
- [101] C. Jentzsch. (Jun. 2019). Blockchains Acquires Slock.it. [Online]. Available: https://blog.slock.it/blockchains-acquires-slock-it-4b3a0276893d
- [102] F. Hawlitschek, B. Notheisen, and T. Teubner, "A 2020 perspective on 'the limits of trust-free systems: A literature review on blockchain technology and trust in the sharing economy," Electron. Commerce Res. Appl., vol. 40, Mar. 2020, Art. no. 100935, doi: 10.1016/j.elerap.2020.100935.
- [103] A. Suliman, Z. Husain, M. Abououf, M. Alblooshi, and K. Salah, "Monetization of IoT data using smart contracts," IET Netw., vol. 8, no. 1, pp. 32–37, Jan. 2019. [104] E. Mengelkamp, J. Gärttner, K. Rock, S. Kessler, L. Orsini, and C. Wein-
- hardt, "Designing microgrid energy markets," Appl. Energy, vol. 210, pp. 870-880, Jan. 2018.
- [105] A. Goranović, M. Meisel, L. Fotiadis, S. Wilker, A. Treytl, and T. Sauter, "Blockchain applications in microgrids an overview of current projects and concepts," in Proc. 43rd Annu. Conf. IEEE Ind. Electron. Soc. (IECON), Oct. 2017, pp. 6153-6158.
- [106] S. Khalili, V. Disfani, and M. Ahmadi, "Impact of blockchain technology on electric power grids-A case study in LO3 energy," Jun. 2021, arXiv:2106.05395.
- [107] B. Cohen and K. Pietrzak, "The Chia network blockchain," Tech. Rep., Jul. 2019, p. 44.
- [108] S. Ezziri and O. Khadir, "A zero-knowledge identification scheme based on the discrete logarithm problem and elliptic curves," in Advances on Smart and Soft Computing. Singapore: Springer, Oct. 2020, pp. 27-35, doi: 10.1007/978-981-15-6048-4 3.
- [109] ConsenSys. (2017). Grid+: Welcome to the Future of Energy. [Online]. Available: https://cryptorating.eu/whitepapers/Grid+/Gridwhitepaper.pdf
- [110] A. S. Musleh, G. Yao, and S. M. Muyeen, "Blockchain applications in smart grid-review and frameworks," IEEE Access, vol. 7, pp. 86746-86757, 2019.
- [111] M. Mihaylov, S. Jurado, N. Avellana, K. Van Moffaert, I. M. de Abril, and A. Nowe, "NRGcoin: Virtual currency for trading of renewable energy in smart grids," in Proc. 11th Int. Conf. Eur. Energy Market (EEM), May 2014, pp. 1-6. [Online]. Available: http://ieeexplore. ieee.org/document/6861213/
- [112] M. Mihaylov, I. Razo-Zapata, and A. Nowé, "NRGcoin-A blockchain-based reward mechanism for both production and consumption of renewable energy," in Transforming Climate Finance and Green Investment With Blockchains. Amsterdam, The Netherlands: Elsevier, 2018, pp. 111-131. [Online]. Available: https://linkinghub.elsevier.com/retrieve/pii/B9780128144473000094
- [113] M. Mihaylov, R. Rădulescu, I. Razo-Zapata, S. Jurado, L. Arco, N. Avellana, and A. Nowé, "Comparing stakeholder incentives across state-of-the-art renewable support mechanisms," Renew. Energy, vol. 131, pp. 689-699, Feb. 2019.
- [114] M. Luke, S. Lee, Z. Pekarek, and A. Dimitrova, "Blockchain in electricity: A critical review of progress to date," NERA Econ. Consulting, New York, NY, USA, Tech. Rep., May 2018.
- [115] F. Vanrykel, D. Ernst, and M. Bourgeois, "Regulatory challenges for share&charge models," Netw. Ind. Quart., vol. 20, no. 3, pp. 17-21, 2018.
- [116] S. Zhu, M. Song, M. K. Lim, J. Wang, and J. Zhao, "The development of energy blockchain and its implications for China's energy sector," Resour. Policy, vol. 66, Jun. 2020, Art. no. 101595, doi: 10.1016/j.resourpol.2020.101595.
- [117] Sonnen. (May 2017). Stabilizing the Power Grid With Households: Tennet and Sonnen are Pioneering the Networking of Storage Batteries With Blockchain Technology. [Online]. Available: https:// sonnenusa.com/en/stabilizing-power-grid-households-tennet-andsonnen-are-pioneering-networking-storage-batteries/

- [118] S. Höhne and V. Tiberius, "Powered by blockchain: Forecasting blockchain use in the electricity market," *Int. J. Energy Sector Manage.*, vol. 14, no. 6, pp. 1221–1238, May 2020, doi: 10.1108/ijesm-10-2019-0002.
- [119] Q. Yang and H. Wang, "Exploring blockchain for the coordination of distributed energy resources," in *Proc. 55th Annu. Conf. Inf. Sci. Syst.* (CISS), Mar. 2021, pp. 1–6.
- [120] A. El-Zonkoly, "Feasibility of blockchain-based energy trading within islanded microgrids in Alexandria, Egypt," J. Energy Eng., vol. 147, no. 3, Jun. 2021, Art. no. 04021009, doi: 10.1061/(asce)ey.1943-7897.0000754.
- [121] Q. Yang and H. Wang, "Privacy-preserving transactive energy management for IoT-aided smart homes via blockchain," *IEEE Internet Things J.*, vol. 8, no. 14, pp. 11463–11475, Jul. 2021, doi: 10.1109/JIOT.2021.3051323.
- [122] L. Bastida, J. J. Cohen, A. Kollmann, A. Moya, and J. Reichl, "Exploring the role of ICT on household behavioural energy efficiency to mitigate global warming," *Renew. Sustain. Energy Rev.*, vol. 103, pp. 455–462, Apr. 2019.
- [123] A. Rubino, A. Sapio, and M. La Scala, Handbook of Energy Economics and Policy: Fundamentals and Applications for Engineers and Energy Planners. Amsterdam, The Netherlands: Elsevier, 2021.
- [124] M. Mansoor and J. Paul, "Impact of energy efficiency-based ICT adoptions on prosumers and consumers," J. Cleaner Prod., vol. 331, Jan. 2022, Art. no. 130008.
- [125] J. L. Mirez, H. R. Chamorro, C. A. Ordonez, and R. Moreno, "Energy management of distributed resources in microgrids," in *Proc. IEEE 5th Colombian Workshop Circuits Syst. (CWCAS)*, Oct. 2014, pp. 1–5.
- [126] L. Edmonds, B. Liu, H. Zhang, C. Scoglio, D. Gruenbacher, and H. Wu, "Blockchain-enabled transactive home energy management systems in distribution networks," in *Proc. IEEE Kansas Power Energy Conf.* (KPEC), Jul. 2020, pp. 1–5.
- [127] S. N. Islam, M. A. Mahmud, and A. M. T. Oo, "A communication scheme for blockchain based peer to peer energy trading," in *Proc. IEEE Power Energy Soc. Gen. Meeting (PESGM)*, Aug. 2020, pp. 1–5.
- [128] B. Couraud, V. Robu, D. Flynn, M. Andoni, S. Norbu, and H. Quinard, "Real-time control of distributed batteries with blockchain-enabled market export commitments," *IEEE Trans. Sustain. Energy*, vol. 13, no. 1, pp. 579–591, Jan. 2022.
- [129] S. N. G. Gourisetti, D. J. Sebastian-Cardenas, B. Bhattarai, P. Wang, S. Widergren, M. Borkum, and A. Randall, "Blockchain smart contract reference framework and program logic architecture for transactive energy systems," *Appl. Energy*, vol. 304, Dec. 2021, Art. no. 117860.
- [130] Q. Yang and H. Wang, "Blockchain-empowered socially optimal transactive energy system: Framework and implementation," *IEEE Trans. Ind. Informat.*, vol. 17, no. 5, pp. 3122–3132, May 2021.
- [131] B. V. Rao, M. Stefan, R. Schwalbe, R. Karl, F. Kupzog, and M. Kozek, "Stratified control applied to a three-phase unbalanced low voltage distribution grid in a local peer-to-peer energy community," *Energies*, vol. 14, no. 11, p. 3290, Jun. 2021.
- [132] J. García-Hernández, L. G. Marín-Collazos, G. Jiménez-Estévez, and P. Mendoza-Araya, "Distributed ledger technologies based microgrid energy management using IOTA tangle," in *Proc. IEEE CHILEAN Conf. Electr., Electron. Eng., Inf. Commun. Technol. (CHILECON)*, Dec. 2021, pp. 1–5.
- pp. 1–5.
 [133] X. Liu, "Research on decentralized operation scheduling strategy of integrated energy system based on energy blockchain," *Int. J. Energy Res.*, Dec. 2021.
- [134] X. Huang, Y. Zhang, D. Li, and L. Han, "An optimal scheduling algorithm for hybrid EV charging scenario using consortium blockchains," *Future Gener. Comput. Syst.*, vol. 91, pp. 555–562, Feb. 2019.
 [135] Y. Li and B. Hu, "An iterative two-layer optimization charging
- [135] Y. Li and B. Hu, "An iterative two-layer optimization charging and discharging trading scheme for electric vehicle using consortium blockchain," *IEEE Trans. Smart Grid*, vol. 11, no. 3, pp. 2627–2637, May 2020.
- [136] Q. Zhang, Y. Su, X. Wu, Y. Zhu, and Y. Hu, "Electricity trade strategy of regional electric vehicle coalitions based on blockchain," *Electr. Power Syst. Res.*, vol. 204, Mar. 2022, Art. no. 107667.
- [137] T. Sousa, T. Soares, P. Pinson, F. Moret, T. Baroche, and E. Sorin, "Peerto-peer and community-based markets: A comprehensive review," *Renew. Sustain. Energy Rev.*, vol. 104, pp. 367–378, Apr. 2019.
- [138] B. Solanki, A. Agarwal, R. Meena, N. Mahiya, D. Sharma, P. Kushwaha, P. Mathuria, and R. Bhakar, "Blockchain-based decentralized hybrid P2P energy trading," in *Proc. 9th IEEE Int. Conf. Power Syst. (ICPS)*, Dec. 2021, pp. 1–5.

- [139] M. M. Esfahani, "A hierarchical blockchain-based electricity market framework for energy transactions in a security-constrained cluster of microgrids," *Int. J. Electr. Power Energy Syst.*, vol. 139, Jul. 2022, Art. no. 108011.
- [140] O. Bouachir, M. Aloqaily, Ö. Özkasap, and F. Ali, "FederatedGrids: Federated learning and blockchain-assisted P2P energy sharing," *IEEE Trans. Green Commun. Netw.*, vol. 6, no. 1, pp. 424–436, Mar. 2022.
- [141] M. Mehdinejad, H. Shayanfar, and B. Mohammadi-Ivatloo, "Decentralized blockchain-based peer-to-peer energy-backed token trading for active prosumers," *Energy*, vol. 244, Apr. 2022, Art. no. 122713.
- [142] M. F. Munoz, K. Zhang, and F. Amara, "ZipZap: A blockchain solution for local energy trading," 2022, *arXiv:2202.13450*.
 [143] O. Samuel, N. Javaid, T. A. Alghamdi, and N. Kumar, "Towards sus-
- [143] O. Samuel, N. Javaid, T. A. Alghamdi, and N. Kumar, "Towards sustainable smart cities: A secure and scalable trading system for residential homes using blockchain and artificial intelligence," *Sustain. Cities Soc.*, vol. 76, Jan. 2022, Art. no. 103371.
- [144] S. Wang, A. F. Taha, J. Wang, K. Kvaternik, and A. Hahn, "Energy crowdsourcing and peer-to-peer energy trading in blockchain-enabled smart grids," *IEEE Trans. Syst., Man, Cybern., Syst.*, vol. 49, no. 8, pp. 1612–1623, Aug. 2019.
- [145] A. Aoun, H. Ibrahim, M. Ghandour, and A. Ilinca, "Blockchain-enabled energy demand side management cap and trade model," *Energies*, vol. 14, no. 24, p. 8600, Dec. 2021.
- [146] C. Dang, J. Zhang, C.-P. Kwong, and L. Li, "Demand side load management for big industrial energy users under blockchain-based peerto-peer electricity market," *IEEE Trans. Smart Grid*, vol. 10, no. 6, pp. 6426–6435, Nov. 2019.
- [147] L. Yinan, Y. Wentao, H. Ping, C. Chang, and W. Xiaonan, "Design and management of a distributed hybrid energy system through smart contract and blockchain," Tech. Rep., 2019.
- [148] Z. A. E. Houda, A. Hafid, and L. Khoukhi, "Blockchain meets AMI: Towards secure advanced metering infrastructures," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2020, pp. 1–6. [Online]. Available: https://ieeexplore.ieee.org/document/9148963/
- [149] M. Mylrea and S. N. G. Gourisetti, "Blockchain for smart grid resilience: Exchanging distributed energy at speed, scale and security," in *Proc. Resilience Week (RWS)*, Sep. 2017, pp. 18–23. [Online]. Available: http://ieeexplore.ieee.org/document/8088642/
- [150] C. Plaza, J. Gil, F. de Chezelles, and K. A. Strang, "Distributed solar selfconsumption and blockchain solar energy exchanges on the public grid within an energy community," in *Proc. IEEE Int. Conf. Environ. Electr. Eng.*, Jun. 2018, pp. 1–4.
- [151] C. Pop, T. Cioara, M. Antal, I. Anghel, I. Salomie, and M. Bertoncini, "Blockchain based decentralized management of demand response programs in smart energy grids," *Sensors*, vol. 18, no. 1, p. 162, Jan. 2018.
- [152] S. Tan, X. Wang, and C. Jiang, "Privacy-preserving energy scheduling for ESCOs based on energy blockchain network," *Energies*, vol. 12, no. 8, p. 1530, Apr. 2019.
- [153] A. K. Erenoğlu, I. Şengör, O. Erdinç, and J. P. Catalão, "Blockchain and its application fields in both power economy and demand side management," in *Blockchain-Based Smart Grids*. Amsterdam, The Netherlands: Elsevier, 2020, pp. 103–130, doi: 10.1016/b978-0-12-817862-1. 00006-3.
- [154] S. Acharya, Y. Dvorkin, and R. Karri, "Public plug-in electric vehicles + grid data: Is a new cyberattack vector viable?" *IEEE Trans. Smart Grid*, vol. 11, no. 6, pp. 5099–5113, Nov. 2020.
- [155] Z. Su, Y. Wang, Q. Xu, M. Fei, Y.-C. Tian, and N. Zhang, "A secure charging scheme for electric vehicles with smart communities in energy blockchain," *IEEE Internet Things J.*, vol. 6, no. 3, pp. 4601–4613, Jun. 2019, doi: 10.1109/JIOT.2018.2869297.
- [156] B. Kirpes and C. Becker, "Processing electric vehicle charging transactions in a blockchain-based information system," in *Proc. Amer. Conf. Inf. Syst. (AMCIS).* Atlanta, GA, USA: AIS, Aug. 2018, p. 5.
- [157] F. Knirsch, A. Unterweger, and D. Engel, "Privacy-preserving blockchain-based electric vehicle charging with dynamic tariff decisions," *Comput. Sci. Res. Develop.*, vol. 33, nos. 1–2, pp. 71–79, Sep. 2017, doi: 10.1007/s00450-017-0348-5.
- [158] X. Huang, C. Xu, P. Wang, and H. Liu, "LNSC: A security model for electric vehicle and charging pile management based on blockchain ecosystem," *IEEE Access*, vol. 6, pp. 13565–13574, 2018.
- [159] S. Jain, N. J. Ahuja, P. Srikanth, K. V. Bhadane, B. Nagaiah, A. Kumar, and C. Konstantinou, "Blockchain and autonomous vehicles: Recent advances and future directions," *IEEE Access*, vol. 9, pp. 130264–130328, 2021.

- [160] Y. Wang, Z. Su, J. Li, N. Zhang, K. Zhang, K.-K.-R. Choo, and Y. Liu, "Blockchain-based secure and cooperative private charging pile sharing services for vehicular networks," *IEEE Trans. Veh. Technol.*, vol. 71, no. 2, pp. 1857–1874, Feb. 2022.
- [161] Y. Wang, Z. Su, and N. Zhang, "BSIS: Blockchain-based secure incentive scheme for energy delivery in vehicular energy network," *IEEE Trans. Ind. Informat.*, vol. 15, no. 6, pp. 3620–3631, Jun. 2019.
 [162] 4NEW. (2018). *The KWATT Coin: Introducing Tokenized*
- [162] 4NEW. (2018). The KWATT Coin: Introducing Tokenized Electricity. [Online]. Available: https://technologyisthemessage.files. wordpress.com/2019/03/4new-whitepaper.pdf
- [163] K. Mannaro, A. Pinna, and M. Marchesi, "Crypto-trading: Blockchainoriented energy market," in *Proc. AEIT Int. Annu. Conf.*, Sep. 2017, pp. 1–5, doi: 10.23919/aeit.2017.8240547.
- [164] D. Bates. (Mar. 2018). ImpactPPA: The world's decentralized energy platform. ImpactPPA. [Online]. Available: https://www.impactppa.com/wpcontent/uploads/2018/03/ImpactPPA_WP_v1.2WEB.pdf
- [165] F. C. Silva, M. A. Ahmed, J. M. Martínez, and Y.-C. Kim, "Design and implementation of a blockchain-based energy trading platform for electric vehicles in smart campus parking lots," *Energies*, vol. 12, no. 24, p. 4814, Dec. 2019, doi: 10.3390/en12244814.
- [166] I. El-Sayed, K. Khan, X. Dominguez, and P. Arboleya, "A real pilotplatform implementation for blockchain-based peer-to-peer energy trading," in *Proc. IEEE Power Energy Soc. Gen. Meeting (PESGM)*, Aug. 2020, pp. 1–5.
- [167] E. Šorin, L. Bobo, and P. Pinson, "Consensus-based approach to peer-topeer electricity markets with product differentiation," *IEEE Trans. Power Syst.*, vol. 34, no. 2, pp. 994–1004, Mar. 2019.
- [168] S. Boyd, N. Parikh, E. Chu, B. Peleato, and J. Eckstein, "Distributed optimization and statistical learning via the alternating direction method of multipliers," *Found. Trends Mach. Learn.*, vol. 3, no. 1, pp. 1–122, Jan. 2010.
- [169] A. J. Conejo, E. Castillo, R. Minguez, and R. Garcia-Bertrand, Decomposition Techniques in Mathematical Programming: Engineering and Science Applications. Berlin, Germany: Springer, 2006.
- [170] S. Seven, G. Yao, A. Soran, A. Onen, and S. M. Muyeen, "Peerto-peer energy trading in virtual power plant based on blockchain smart contracts," *IEEE Access*, vol. 8, pp. 175713–175726, 2020, doi: 10.1109/ACCESS.2020.3026180.
- [171] U. Halden, U. Cali, M. F. Dynge, J. Stekli, and L. Bai, "DLT-based equity crowdfunding on the techno-economic feasibility of solar energy investments," *Sol. Energy*, vol. 227, pp. 137–150, Oct. 2021.
- [172] U. Cali, U. Halden, M. F. Dynge, and A.-S. Bukvic-Schaefer, "Blockchain-enabled equity crowdfunding for energy storage investments," in *Proc. Int. Conf. Smart Energy Syst. Technol. (SEST)*, Sep. 2021, pp. 1–6.
- [173] SunExchange. SunExchange: The Global Solar Cell Leasing Platform. Accessed: Nov. 6, 2021. [Online]. Available: https://thesunexchange. com/how-it-works
- [174] Equigy: A Multi-TSO Initiative. Accessed: Nov. 6, 2021. [Online]. Available: https://equigy.com/wp-content/uploads/2020/09/Equigy-Amulti-TSO-initiative.pdf
- [175] Unergy. Unergy S.A.S. Accessed: Mar. 20, 2021. [Online]. Available: https://unergy.io
- [176] G. van Leeuwen, T. AlSkaif, M. Gibescu, and W. van Sark, "An integrated blockchain-based energy management platform with bilateral trading for microgrid communities," *Appl. Energy*, vol. 263, Apr. 2020, Art. no. 114613.
- [177] H. M. Rouzbahani, H. Karimipour, and L. Lei, "A review on virtual power plant for energy management," *Sustain. Energy Technol. Assessments*, vol. 47, Oct. 2021, Art. no. 101370.
- [178] C. DeCusatis and K. Lotay, "Secure, decentralized energy resource management using the ethereum blockchain," in *Proc. 17th IEEE Int. Conf. Trust, Secur. Privacy Comput. Commun./12th IEEE Int. Conf. Big Data Sci. Eng. (TrustCom/BigDataSE)*, Aug. 2018, pp. 1907–1913.
- [179] J. I. Guerrero, E. Personal, A. García, A. Parejo, F. Pérez, and C. León, "Distributed charging prioritization methodology based on evolutionary computation and virtual power plants to integrate electric vehicle fleets on smart grids," *Energies*, vol. 12, no. 12, p. 2402, Jun. 2019.
- [180] A. Mnatsakanyan, H. Albeshr, A. A. Marzooqi, and E. Bilbao, "Blockchain-integrated virtual power plant demonstration," in *Proc.* 2nd Int. Conf. Smart Power Internet Energy Syst. (SPIES), Sep. 2020, pp. 172–175.
 [181] Z. Wang, M. Ogbodo, H. Huang, C. Qiu, M. Hisada, and A. B. Abdallah,
- [181] Z. Wang, M. Ogbodo, H. Huang, C. Qiu, M. Hisada, and A. B. Abdallah, "AEBIS: AI-enabled blockchain-based electric vehicle integration system for power management in smart grid platform," *IEEE Access*, vol. 8, pp. 226409–226421, 2020.

- [182] M. U. Hassan, M. H. Rehmani, and J. Chen, "VPT: Privacy preserving energy trading and block mining mechanism for blockchain based virtual power plants," 2021, arXiv:2102.01480.
- [183] M. Yalpanian, N. Mirzaei, A. Parvizimosaed, F. Farmani, M. Parvizimosaed, and B. Bahrami, "BIOT: A blockchain-based IoT platform for distributed energy resource management," in *Silicon Valley Cybersecurity Conference*. Cham, Switzerland: Springer, 2020, pp. 134–147.
- pp. 134–147.
 [184] J. Zhao, C. He, C. Peng, and X. Zhang, "Blockchain for effective renewable energy management in the intelligent transportation system," *J. Interconnection Netw.*, Jul. 2021, Art. no. 2141009.
- [185] A. Farao, C. Ntantogian, C. Istrate, G. Suciu, and C. Xenakis, "Sealed-GRID: Scalable, trusted, and interoperable pLatform for secured smart GRID," in *Proc. 6th Int. Symp. ICS SCADA Cyber Secur. Res.*, Sep. 2019, pp. 74–81.
- [186] WeCharg. Charg Coin Whitepaper. Accessed: Jul. 29, 2021. [Online]. Available: https://chgcoin.org/white-paper/
- [187] L. P. Neves and F. Conti, "Emerging technologies applied to the renewable energy sector," Konrad Adenauer Stiftung FGV Int. Intell. Unit, Berlin, Germany, Tech. Rep., 2019.
- [188] CUBi Energia. CUBi—Monitoramento e Gestão de Energia Elétrica Para Empresas. Accessed: Aug. 1, 2021. [Online]. Available: https://www.cubienergia.com/
- [189] Dexentralize. (Jan. 2020). Les Presentamos la Plataforma de Comercialización de Energía Dexentralize. [Online]. Available: https://dexentralize.io/
- [190] Efforce. Efforce: The First Blockchain-Based Energy-Saving Platform. Accessed: Jan. 27, 2021. [Online]. Available: https://efforce.io/project
 [191] S. Wozniak, J. Visetti, J. Vanetti, K. Hardesty, S. Scozzese, and
- [191] S. Wozniak, J. Visetti, J. Vanetti, K. Hardesty, S. Scozzese, and A. Castiglione, "Efforce; energy efficiency reinvented," Efforce, Valletta, Malta, Tech. Rep., 2020.
- [192] G. George, R. K. Merrill, and S. J. D. Schillebeeckx, "Digital sustainability and entrepreneurship: How digital innovations are helping tackle climate change and sustainable development," *Entrepreneurship Theory Pract.*, vol. 45, Jan. 2020, Art. no. 104225871989942, doi: 10.1177/1042258719899425.
- [193] J. Lin, M. Pipattanasomporn, and S. Rahman, "Comparative analysis of auction mechanisms and bidding strategies for P2P solar transactive energy markets," *Appl. Energy*, vol. 255, Dec. 2019, Art. no. 113687, doi: 10.1016/j.apenergy.2019.113687.
 [194] S. You and H. Cai, "Early experience of the energy internet: A review
- [194] S. You and H. Cai, "Early experience of the energy internet: A review of demonstrations and pilot applications in Europe," in *Energy Internet*. Cham, Switzerland: Springer, 2020, pp. 121–133, doi: 10.1007/978-3-030-45453-1_4.
- [195] R. Chitchyan and J. Murkin, "Review of blockchain technology and its expectations: Case of the energy sector," 2018, arXiv:1803.03567.
- [196] P. Spring, L. Dijkstra, A. De Almeida, and R. Vrees. (2020). The green light: Enpowering participation in the energy transition with blockchain. IBM Blockchain. [Online]. Available: https://www. ibm.com/downloads/cas/KABXYGAR
- [197] The Energy Web Chain Accelerating the Energy Transition With an Open-Source, Decentralized Blockchain Platform, Energy Web Found., Berlin, Germany, Jul. 2019.
- [198] A. Trbovich, S. Hambridge, D. van den Biggelaar, E. Hesse, and F. Sioshansi, "D3A energy exchange for a transactive grid," in *Behind and Beyond the Meter*. Amsterdam, The Netherlands: Elsevier, 2020, pp. 267–284, doi: 10.1016/b978-0-12-819951-0. 00013-x.
- [199] M. Pichler, M. Meisel, A. Goranovic, K. Leonhartsberger, G. Lettner, G. Chasparis, H. Vallant, S. Marksteiner, and H. Bieser, "Decentralized energy networks based on blockchain: Background, overview and concept discussion," in *Proc. Int. Conf. Bus. Inf. Syst.* Cham, Switzerland: Springer, 2018, pp. 244–257.
- [200] A. Bumpus. (May 2019). Enabling digital energy entrepreneurship: A comparative analysis of German and Australian digital energy entrepreneurial opportunities. Energy Transition Hub. [Online]. Available: https://www.energy-transition-hub.org/files/resource/ attachment/digital_energy_entrepreneurship.pdf
 [201] M. Foti and M. Vavalis, "What blockchain can do for power grids?"
- [201] M. Foti and M. Vavalis, "What blockchain can do for power grids?" Blockchain, Res. Appl., vol. 2, no. 1, Mar. 2021, Art. no. 100008, doi: 10.1016/j.bcra.2021.100008.
- [202] J. Schlund, L. Ammon, and R. German, "ETHome: Open-source blockchain based energy community controller," in *Proc. 9th Int. Conf. Future Energy Syst.*, Jun. 2018, pp. 319–323. [Online]. Available: https://dl.acm.org/doi/10.1145/3208903.3208929

- [203] J. Kosowatz, "Blockchain enables new power markets," *Mech. Eng.*, vol. 141, no. 6, pp. 26–31, Jun. 2019, doi: 10.1115/1.2019-jun1.
- [204] J. Radl, A. Fleischhacker, F. H. Revheim, G. Lettner, and H. Auer, "Comparison of profitability of PV electricity sharing in renewable energy communities in selected European countries," *Energies*, vol. 13, no. 19, p. 5007, Sep. 2020, doi: 10.3390/en13195007.
- [205] Z. Arifin, "The effects of disruptive technologies on power utility company's performance; empirical evidence from Indonesia," *Technol. Anal. Strategic Manage.*, vol. 34, no. 4, pp. 461–473, 2022, doi: 10.1080/09537325.2021.1906853.
- [206] Greeneum Net. (Nov. 2017). Greeneum Global Energy Networks Whitepaper. [Online]. Available: https://static1.squarespace.com/static/ 5a3de333c027d8dd95183ca7/t/5a750b678165f51a34ddeb6e/151762007 5918/Greeneum+Whitepaper+v2.pdf
- [207] Lition White Paper, the Blockchain Standard Infrastructure for Business. Accessed: Jun. 20, 2021. [Online]. Available: https://lition. io/docs/Lition_Whitepaper.pdf
- [208] D. Livingston, V. Sivaran, M. Freeman, and M. Fiege, "Applying blockchain technology to electric power systems," *JSTOR*, pp. 1–35, Jul. 2018.
- [209] A. M. Liceaga, P. Sanchis, and M. D. M. Rubio-Varas, "New business models as drivers of distributed renewable energy systems," in *Proc. IEEE Int. Conf. Environ. Electr. Eng. IEEE Ind. Commercial Power Syst. Eur. (EEEIC/ICPS Europe)*, Jun. 2018, pp. 1–6.
- [210] P. Fairley, "Swarm electrification' powers villages," *IEEE Spectr.*, vol. 55, no. 4, p. 21, Apr. 2018.
- [211] P. Bronder, "Effective disruption: How blockchain technology can transform the energy sector," in *Sustainable Innovation and Impact*. Evanston, IL, USA: Routledge, 2018, pp. 208–212.
- [212] MyBit. Accessed: Nov. 7, 2021. [Online]. Available: https://mybit.io/
- [213] Power2Peer. (2018). Power2Peer: Blockchain Innovation for Energy Independence. [Online]. Available: https://power2peer.com/wpcontent/uploads/2018/01/power2peer_executivesummary.pdf
- [214] D. Chen, "Utility of the blockchain for climate mitigation," J. Brit. Blockchain Assoc., vol. 1, no. 1, p. 3577, 2018.
- [215] L.-G. Vandenberg. (Jun. 2018). Solara.io: Clean energy blockchain network. Solara Australia Limited. [Online]. Available: https://solara.io/ whitepaper/
- [216] SolarCoin Foundation. SolarCoin. Accessed: Feb. 15, 2021. [Online]. Available: https://solarcoin.org/
- [217] L. P. Johnson, A. Islam, N. Gogerty, and J. Zitoli, "Connecting the blockchain to the sun to save the planet," *SSRN Electron. J.*, pp. 1–16, Dec. 2015, doi: 10.2139/ssrn.2702639.
- [218] Z. Nehai and G. Guérard, "Integration of the blockchain in a smart grid model," in *Proc. 14th Int. Conf. Young Scientists Energy Issues (CYSENI)*, 2017, pp. 127–134.
- [219] J. Cuenca, E. Jamil, and B. Hayes, "Energy communities and sharing economy concepts in the electricity sector: A survey," in *Proc. IEEE Int. Conf. Environ. Electr. Eng. IEEE Ind. Commercial Power Syst. Eur.* (*EEEIC/ICPS Europe*), Jun. 2020, pp. 1–6.
- [220] P. Wongthongtham, D. Marrable, B. Abu-Salih, X. Liu, and G. Morrison, "Blockchain-enabled peer-to-peer energy trading," *Comput. Electr. Eng.*, vol. 94, Sep. 2021, Art. no. 107299, doi: 10.1016/j.compeleceng.2021.107299.
- [221] Synergy Electrify: Asia's Electric Marketplace 2021. Accessed: Jul. 23, 2021. [Online]. Available: https://electrify.asia/synergy/
 [222] J. M. España and J. P. Cardenas, "Transactive energy Colombia initia-
- [222] J. M. España and J. P. Cardenas, "Transactive energy Colombia initiative," Tech. Rep., Aug. 2019, p. 20.
- [223] EnergEIA. Transactive Energy Colombia. Accessed: Feb. 4, 2021. [Online]. Available: http://www.transactive-energy.co
- [224] S. Annala, L. Klein, L. Matos, S. Repo, O. Kilkki, A. Narayanan, and S. Honkapuro, "Framework to facilitate electricity and flexibility trading within, to, and from local markets," *Energies*, vol. 14, no. 11, p. 3229, May 2021, doi: 10.3390/en14113229.
- [225] Verv VLUX White Paper—The Evolution of Energy. Accessed: Nov. 7, 2021. [Online]. Available: https://vlux.io/media/VLUX_ Whitepaper.pdf
- [226] A. Rot, A. Varnavskiy, U. Gruzina, V. Trubnikov, A. Buryakova, and E. Sebechenko, "Design of models for the tokenization of electric power industry basing on the blockchain technology," in *Proc. Commun. Papers Federated Conf. Comput. Sci. Inf. Syst.*, Sep. 2018, pp. 45–50, doi: 10.15439/2018f330.
- [227] N. Martyniuk, K. Kaarlep, K. Kuhi, J. Tuisk, and M. John. (Feb. 2019). WePower whitepaper version 2. WePower Team. [Online]. Available: https://wepower.com/media/WhitePaper-WePower_v_2.pdf

- [228] Z. Liu, D. Wang, J. Wang, X. Wang, and H. Li, "A blockchain-enabled secure power trading mechanism for smart grid employing wireless networks," *IEEE Access*, vol. 8, pp. 177745–177756, 2020.
- [229] M. Aloqaily, O. Bouachir, Ö. Özkasap, and F. S. Ali, "SynergyGrids: Blockchain-supported distributed microgrid energy trading," *Peer-to-Peer Netw. Appl.*, vol. 15, pp. 884–900, Mar. 2022.
- [230] X. Sun, S. Yan, N. Yang, Z. Ding, C. Shen, and Z. Zhong, "Downlink NOMA transmission for low-latency short-packet communications," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, May 2018, pp. 1–6.
- pp. 1–6.
 [231] M. Gough, S. F. Santos, A. Almeida, M. Lotfi, M. S. Javadi, D. Z. Fitiwi, G. J. Osorio, R. Castro, and J. P. S. Catalao, "Blockchain-based transactive energy framework for connected virtual power plants," *IEEE Trans. Ind. Appl.*, vol. 58, no. 1, pp. 986–995, Jan. 2022.
 [232] D. Efanov and P. Roschin, "The all-pervasiveness of the blockchain
- [232] D. Efanov and P. Roschin, "The all-pervasiveness of the blockchain technology," *Proc. Comput. Sci.*, vol. 123, pp. 116–121, Jan. 2018.
- [233] S. Maitra, V. P. Yanambaka, D. Puthal, A. Abdelgawad, and K. Yelamarthi, "Integration of Internet of Things and blockchain toward portability and low-energy consumption," *Trans. Emerg. Telecommun. Technol.*, vol. 32, no. 6, Jun. 2021, Art. no. e4103.
- [234] N. Radziwill, "Blockchain revolution: How the technology behind bitcoin is changing money, business, and the world," *Qual. Manage. J.*, vol. 25, no. 1, pp. 64–65, Jan. 2018.
- [235] E. Mengelkamp, B. Notheisen, C. Beer, D. Dauer, and C. Weinhardt, "A blockchain-based smart grid: Towards sustainable local energy markets," *Comput. Sci.-Res. Develop.*, vol. 33, nos. 1–2, pp. 207–214, 2018Feb. 2018, doi: 10.1007/s00450-017-0360-9.
- [236] M. Aybar-Mejía, D. Rosario-Weeks, D. Mariano-Hernández, and M. Domínguez-Garabitos, "An approach for applying blockchain technology in centralized electricity markets," *Electr. J.*, vol. 34, no. 3, Apr. 2021, Art. no. 106918.
- [237] C. Pike and A. Capobianco, "Antitrust and the trust machine," in Proc. OECD Blockchain Policy Ser., Jan. 2020, pp. 1–20.
- [238] A. Hasankhani, S. M. Hakimi, M. Bisheh-Niasar, M. Shafie-Khah, and H. Asadolahi, "Blockchain technology in the future smart grids: A comprehensive review and frameworks," *Int. J. Electr. Power Energy Syst.*, vol. 129, Jul. 2021, Art. no. 106811, doi: 10.1016/j.ijepes.2021.106811.
- [239] I. Zografopoulos, J. Ospina, X. Liu, and C. Konstantinou, "Cyberphysical energy systems security: Threat modeling, risk assessment, resources, metrics, and case studies," *IEEE Access*, vol. 9, pp. 29775–29818, 2021.
- [240] U. Cali and C. Lima, "Energy informatics using the distributed ledger technology and advanced data analytics," in *Cases on Green Energy and Sustainable Development*. Hershey, PA, USA: IGI Global, 2020, pp. 438–481.
 [241] Y. Sang, U. Cali, M. Kuzlu, M. Pipattanasomporn, C. Lima, and S. Chen,
- [241] Y. Sang, U. Cali, M. Kuzlu, M. Pipattanasomporn, C. Lima, and S. Chen, "IEEE SA blockchain in energy standardization framework: Grid and prosumer use cases," in *Proc. IEEE Power Energy Soc. Gen. Meeting* (*PESGM*), Aug. 2020, pp. 1–5.
- [242] S. N. G. Gourisetti, Ü. Ĉali, K.-K.-R. Choo, E. Escobar, C. Gorog, A. Lee, C. Lima, M. Mylrea, M. Pasetti, F. Rahimi, R. Reddi, and A. S. Sani, "Standardization of the distributed ledger technology cybersecurity stack for power and energy applications," *Sustain. Energy, Grids Netw.*, vol. 28, Dec. 2021, Art. no. 100553.
- [243] A. Yagmur, B. A. Dedeturk, A. Soran, J. Jung, and A. Onen, "Blockchainbased energy applications: The DSO perspective," *IEEE Access*, vol. 9, pp. 145605–145625, 2021.
- [244] P. Ge, F. Teng, C. Konstantinou, and S. Hu, "A resilience-oriented centralised-to-decentralised framework for networked microgrids management," *Appl. Energy*, vol. 308, Feb. 2022, Art. no. 118234. [Online]. Available: https://www.sciencedirect.com/science/article/ pii/S0306261921014987
- [245] C. Konstantinou and O. M. Anubi, "Resilient cyber-physical energy systems using prior information based on Gaussian process," *IEEE Trans. Ind. Informat.*, vol. 18, no. 3, pp. 2160–2168, Mar. 2022.
- [246] M. K. Nallapaneni and S. S. Chopra, "Enhancing the resilience of urban networked community microgrids: Blockchain-enabled flexible energy trading strategy," in *Proc. Actionable Sci. Urban Sustain. (AScUS)*, 2020.
- [247] S. Mirsaeidi, S. Li, S. Devkota, M. Li, J. He, D. Tzelepis, C. Konstantinou, and D. M. Said, "A power loss minimization strategy based on optimal placement and sizing of distributed energy resources," *Int. J. Numer. Model., Electron. Netw., Devices Fields*, vol. 35, no. 4, Jul. 2022, Art. no. e3000. [Online]. Available: https://onlinelibrary.wiley.com/doi/abs/10.1002/jnm.3000

...